DOI 10.4171/JEMS/707



J. William Helton · Igor Klep · Scott McCullough

# The tracial Hahn–Banach theorem, polar duals, matrix convex sets, and projections of free spectrahedra

Received August 4, 2014

**Abstract.** This article investigates matrix convex sets and introduces their tracial analogs which we call contractively tracial convex sets. In both contexts completely positive (cp) maps play a central role: unital cp maps in the case of matrix convex sets and trace preserving cp (CPTP) maps in the case of contractively tracial convex sets. CPTP maps, also known as quantum channels, are fundamental objects in quantum information theory.

Free convexity is intimately connected with Linear Matrix Inequalities (LMIs)  $L(x) = A_0 + A_1x_1 + \cdots + A_gx_g \ge 0$  and their matrix convex solution sets  $\{X : L(X) \ge 0\}$ , called *free spectrahedra*. The Effros–Winkler Hahn–Banach Separation Theorem for matrix convex sets states that matrix convex sets are solution sets of LMIs with operator coefficients. Motivated in part by cp interpolation problems, we develop the foundations of convex analysis and duality in the tracial setting, including tracial analogs of the Effros–Winkler Theorem.

The projection of a free spectrahedron in g + h variables to g variables is a matrix convex set called a *free spectrahedrop*. As a class, free spectrahedrops are more general than free spectrahedra, but at the same time more tractable than general matrix convex sets. Moreover, many matrix convex sets can be approximated from above by free spectrahedrops. Here a number of fundamental results for spectrahedrops and their polar duals are established. For example, the free polar dual of a free spectrahedrop is again a free spectrahedrop. We also give a Positivstellensatz for free polynomials that are positive on a free spectrahedrop.

**Keywords.** Linear matrix inequality (LMI), polar dual, LMI domain, spectrahedron, spectrahedrop, convex hull, free real algebraic geometry, noncommutative polynomial, cp interpolation, quantum channel, tracial hull, tracial Hahn–Banach theorem

## Contents

1.	Intro	duction $\ldots$ $\ldots$ $\ldots$ $\ldots$ $184$	-6
	1.1.	Results on polar duals and free spectrahedrops	8
	1.2.	Results on interpolation of cp maps and quantum channels	8

J. W. Helton: Department of Mathematics, University of California, San Diego, CA 92093, USA; e-mail: helton@math.ucsd.edu

I. Klep: Department of Mathematics, The University of Auckland, Auckland 1010, New Zealand; e-mail: igor.klep@auckland.ac.nz

S. McCullough: Department of Mathematics, University of Florida, Gainesville, FL 32611-8105, USA; e-mail: sam@math.ufl.edu

Mathematics Subject Classification (2010): Primary 14P10, 47L25, 90C22; Secondary 13J30, 46L07

	1.3.	Free tracial Hahn–Banach Theorem	9	
	1.4.	Reader's guide	0	
2.		iminaries	0	
		Free sets	0	
	2.2.		1	
	2.3.	Free semialgebraic sets	1	
	2.4.	Free convexity	2	
	2.5.		3	
3.	Com	pletely positive interpolation	4	
		Basics of completely positive maps	4	
		Quantum interpolation problems and semidefinite programming	7	
4.		spectrahedrops and polar duals	8	
		Projections of free spectrahedra: free spectrahedrops	8	
		Basics of polar duals	9	
	4.3.	Polar duals of free spectrahedra	1	
	4.4.	The polar dual of a free spectrahedrop is a free spectrahedrop	4	
	4.5.		0	
5.	Posi	tivstellensatz for free spectrahedrops	0	
		Proof of Theorem 5.1	1	
6.	Trac	ial sets	4	
	6.1.		5	
	6.2.	Contractively tracial sets and hulls	7	
	6.3.	Classical duals of free convex hulls and of tracial hulls	8	
7.	Trac	ial spectrahedra and an Effros–Winkler separation theorem	9	
	7.1.	Tracial spectrahedra	9	
	7.2.	An auxiliary result	0	
	7.3.		1	
	7.4.		4	
	7.5.	Matrix convex tracial sets and free cones	8	
8.		mples	0	
References				
Index				

## 1. Introduction

This article investigates matrix convex sets from the perspective of the emerging areas of free real algebraic geometry and free analysis [Voi04, Voi10, KVV14, MS11, Pop10, AM15, BB07, dOHMP09, HKM13b, PNA10]. It also introduces *contractively tracial convex sets*, the tracial analogs of matrix convex sets appropriate for the quantum channel and quantum operation interpolation problems. Matrix convex sets arise naturally in a number of contexts, including engineering systems theory, operator spaces, systems and algebras, and are inextricably linked to unital completely positive (ucp) maps [SIG97, Arv72, Pau02, Far00, HMPV09]. On the other hand, completely positive trace preserving (CPTP) maps are central to quantum information theory [NC10, JKPP11]. Hence there is an inherent similarity between matrix convex sets and structures naturally occurring in quantum information theory.

Given positive integers g and n, let  $\mathbb{S}_n^g$  denote the set of g-tuples  $X = (X_1, \dots, X_g)$  of complex  $n \times n$  hermitian matrices and let  $\mathbb{S}^g$  denote the sequence  $(\mathbb{S}_n^g)_n$ . We use  $M_n$  to

denote the algebra of  $n \times n$  complex matrices. A subset  $\Gamma \subseteq \mathbb{S}^g$  is a sequence  $\Gamma = (\Gamma(n))_n$ such that  $\Gamma(n) \subseteq \mathbb{S}_n^g$  for each n. A *matrix convex set* is a subset  $\Gamma \subseteq \mathbb{S}^g$  that is closed with respect to direct sums and (simultaneous) conjugation by isometries. *Closed under direct sums* means that if  $X \in \Gamma(n)$  and  $Y \in \Gamma(m)$ , then

$$X \oplus Y := \left( \begin{pmatrix} X_1 & 0\\ 0 & Y_1 \end{pmatrix}, \dots, \begin{pmatrix} X_g & 0\\ 0 & Y_g \end{pmatrix} \right) \in \Gamma(n+m).$$
(1.1)

Likewise, *closed under conjugation by isometries* means that if  $X \in \Gamma(n)$  and V is an  $n \times m$  isometry, then

$$V^*XV := (V^*X_1V, \dots, V^*X_gV) \in \Gamma(m).$$

The simplest examples of matrix convex sets arise as solution sets of linear matrix inequalities (LMIs). The use of LMIs is a major advance in systems engineering in the past two decades [SIG97]. Furthermore, LMIs underlie the theory of semidefinite programming [BPR13, BN02], itself a recent major innovation in convex optimization [Nem06].

Matrix convex sets determined by LMIs are based on a free analog of an affine linear functional, often called a *linear pencil*. Given a positive integer d and g hermitian  $d \times d$  matrices  $A_j$ , let

$$L(x) = A_0 + \sum_{j=1}^{g} A_j x_j.$$
(1.2)

This linear pencil is often denoted by  $L_A$  to emphasize the dependence on A. In the case that  $A_0 = I_d$ , we call L monic. Replacing  $x \in \mathbb{R}^g$  with a tuple  $X = (X_1, \ldots, X_g)$  of  $n \times n$  hermitian matrices and letting  $W \otimes Z$  denote the Kronecker product of matrices leads to the evaluation of the free affine linear functional,

$$L(X) = A_0 \otimes I_n + \sum A_j \otimes X_j.$$
(1.3)

The inequality  $L(X) \succeq 0$  is a *free linear matrix inequality* (*free LMI*). The solution set  $\Gamma$  of this free LMI is the sequence of sets

$$\Gamma(n) = \{ X \in \mathbb{S}_n^g : L(X) \succeq 0 \},\$$

known as a *free spectrahedron* (or *free LMI domain*). It is easy to see that  $\Gamma$  is a matrix convex set.

By the Effros–Winkler matricial Hahn–Banach Separation Theorem [EW97], (up to a technical hypothesis) every matrix convex set is the solution set of  $L(X) \geq 0$ , as in equation (1.3), for some monic linear pencil *provided* the  $A_j$  are allowed to be hermitian operators on a (common) Hilbert space. More precisely, every matrix convex set is a (perhaps infinite) intersection of free spectrahedra. Thus, being a spectrahedron imposes a strict finiteness condition on a matrix convex set.

In between (closed) matrix convex sets and spectrahedra lie the class of domains we call *spectrahedrops*, namely coordinate projections of free spectrahedra. A subset  $\Delta \subseteq \mathbb{S}^{g}$  is a *free spectrahedrop* if there exists a pencil

$$L(x, y) = A_0 + \sum_{j=1}^{g} A_j x_j + \sum_{k=1}^{h} B_k y_k$$

in g + h variables such that

$$\Delta(n) = \{ X \in \mathbb{S}_n^g : \exists Y \in \mathbb{S}_n^h, \ L(X, Y) \succeq 0 \}.$$

$$(1.4)$$

In applications, presented with a convex set, one would like, for optimization purposes say, to know if it is a spectrahedron or a spectrahedrop. Alternatively, presented with an algebraically defined set  $\Gamma \subseteq \mathbb{S}^g$  that is not necessarily convex, it is natural to consider the relaxation obtained by replacing  $\Gamma$  with its matrix convex hull or an approximation thereof. Thus, it is of interest to know when the convex hull of a set is a spectrahedron or perhaps a spectrahedrop. An approach to these problems via approximating from above by spectrahedrops was pursued in [HKM16]. Here we develop the duality approach. Typically the second polar dual of a set is its closed matrix convex hull.

#### 1.1. Results on polar duals and free spectrahedrops

We list here our main results on free spectrahedrops and polar duals. For the reader unfamiliar with the terminology, the definitions not already introduced can be found in Section 2 with the exception of *free polar dual* whose definition appears in Subsection 4.2.

(1) A perfect free Positivstellensatz (Theorem 5.1) for any symmetric free polynomial *p* on a free spectrahedrop Δ as in (1.4) says that *p*(*X*) is positive semidefinite for all *X* ∈ Δ if and only if *p* has the form

$$p(x) = f(x)^* f(x) + \sum_{\ell} q_{\ell}(x)^* L(x, y) q_{\ell}(x)$$

where f and and  $q_{\ell}$  are vectors with polynomial entries. If the degree of p is less than or equal to 2r + 1, then f and  $q_{\ell}$  have degree no greater than r.

- (2) The free polar dual of a free spectrahedrop is a free spectrahedrop (Theorem 4.11 and Corollary 4.17).
- (3) The matrix convex hull of a union of finitely many bounded free spectrahedrops is a bounded free spectrahedrop (Proposition 4.18).
- (4) A matrix convex set is, in a canonical sense, generated by a finite set (equivalently a single point) if and only if it is the polar dual of a free spectrahedron (Theorem 4.6).

#### 1.2. Results on interpolation of cp maps and quantum channels

A completely positive (*cp*) map  $M_n \rightarrow M_m$  that is trace preserving is called a *quantum channel*, and a cp map that is trace nonincreasing for positive semidefinite arguments is a *quantum operation*. These maps figure prominently in quantum information theory [NC10].

The cp interpolation problem is formulated as follows. Given  $A \in \mathbb{S}_n^g$  and  $B \in \mathbb{S}_m^g$ , does there exist a cp map  $\Phi : M_n \to M_m$  such that, for  $1 \le \ell \le g$ ,

$$\Phi(A_\ell) = B_\ell?$$

One can require further that  $\Phi$  be unital, a quantum channel or a quantum operation. Imposing either of the latter two constraints pertains to quantum information theory [Hal1, Kle07, NCSB98], where one is interested in quantum channels (resp., quantum operations) that send a prescribed set of quantum states into another set of quantum states.

*1.2.1. Algorithmic aspects.* A byproduct of the methods used in this paper and in [HKM13a] produces solutions to these cp interpolation problems in the form of an algorithm, Theorem 3.4 in Subsection 3.2. Ambrozie and Gheondea [AG15] solved these interpolation problems with LMI algorithms. While equivalent to theirs, our solutions are formulated as concrete LMIs that can be solved with a standard semidefinite programming (SDP) solver. These interpolation results are a basis for proofs of the results outlined in Section 1.1.

#### 1.3. Free tracial Hahn-Banach Theorem

Matrix convex sets are closely connected with ranges of unital cp maps. Indeed, given a tuple  $A \in \mathbb{S}_m^g$ , the matrix convex hull of the set  $\{A\}$  is the sequence of sets

$$(\{B \in \mathbb{S}_n^g : B_j = \Phi(A_j) \text{ for some ucp map } \Phi : M_m \to M_n\})_n$$

From the point of view of quantum information theory it is natural to consider hulls of ranges of quantum operations. We say that  $\mathcal{Y} \subseteq \mathbb{S}^g$  is *contractively tracial* if for all positive integers m, n and finite collections  $\{C_\ell\}$  of  $n \times m$  matrices such that

$$\sum C_{\ell}^* C_{\ell} \preceq I_m,$$

we have  $\sum C_j Y C_j^* \in \mathcal{Y}(n)$  for all  $Y \in \mathcal{Y}(m)$ . It is clear that an intersection of contractively tracial sets is again contractively tracial, giving rise, in the usual way, to the notion of the *contractive tracial hull*, denoted cthull. For a tuple *A*,

$$\operatorname{cthull}(A) = \{B : \Phi(A) = B \text{ for some quantum operation } \Phi\}.$$

While the unital and quantum interpolation problems have very similar formulations, contractive tracial hulls possess far less structure than matrix convex hulls. A subset  $\mathscr{Y} \subseteq \mathbb{S}^g$ is *levelwise convex* if each  $\mathscr{Y}(m)$  is convex (as a subset of  $\mathbb{S}^g_m$ ). (Generally *levelwise* refers to a property holding for each  $\mathscr{Y}(m) \subseteq \mathbb{S}^g_m$ .) As is easily seen, contractive tracial hulls need not be levelwise convex nor closed with respect to direct sums. However, they do have a few good properties. These we develop in Section 6.

Section 7 contains notions of free spectrahedra and corresponding Hahn–Banach type separation theorems tailored to the tracial setting. To understand *convex* contractively tracial sets, given  $B \in \mathbb{S}_k^g$ , let  $\mathfrak{H}_B = (\mathfrak{H}_B(m))_m$  denote the sequence of sets

$$\mathfrak{H}_B(m) = \left\{ Y \in \mathbb{S}_m^g : \exists T \succeq 0, \ \mathrm{tr}(T) \le 1, \ I \otimes T - \sum B_j \otimes Y_j \succeq 0 \right\}.$$

We call  $\mathfrak{H}_B$  a *tracial spectrahedron*. (Note that  $\mathfrak{H}_B$  is not closed under direct sums, and thus it is not a matrix convex set.) These  $\mathfrak{H}_B$  are all contractively tracial and levelwise

convex. Indeed for such structural reasons, and in view of the tracial Hahn–Banach separation theorem immediately below, we believe these to be the natural analogs of free spectrahedra in the tracial context.

**Theorem 1.1** (cf. Theorem 7.6). If  $\mathcal{Y} \subseteq \mathbb{S}^g$  is contractively tracial, levelwise convex and closed, and if  $Z \in \mathbb{S}_m^g$  is not in  $\mathcal{Y}(m)$ , then there exists a  $B \in \mathbb{S}_m^g$  such that  $\mathcal{Y} \subseteq \mathfrak{H}_B$ , but  $Z \notin \mathfrak{H}_B$ .

Because of the asymmetry between *B* and *Y* in the definition of  $\mathfrak{H}_B$ , there is a second type of tracial spectrahedron. Given  $Y \in \mathbb{S}_k^g$ , we define the *opp-tracial spectrahedron* as the sequence  $\mathfrak{H}_V^{\text{opp}} = (\mathfrak{H}_V^{\text{opp}}(m))_m$  where

$$\mathfrak{H}_{Y}^{\mathrm{opp}}(m) = \Big\{ B \in \mathbb{S}_{m}^{g} : \exists T \geq 0, \ \mathrm{tr}(T) \leq 1, \ I \otimes T - \sum B_{j} \otimes Y_{j} \geq 0 \Big\}.$$
(1.5)

Proposition 7.11 computes the hulls resulting from the two different double duals determined by the two notions of tracial spectrahedron.

#### 1.4. Reader's guide

The paper is organized as follows. Section 2 introduces terminology and notation used throughout the paper. Section 3 solves the cp interpolation problems, and includes a background section on cp maps. Section 4 contains our main results on polar duals, free spectrahedra and free spectrahedrops. It uses the results of Section 3. In particular, we show that a matrix convex set is finitely generated if and only if it is the polar dual of a free spectrahedron (Theorem 4.6). Furthermore, we prove that the polar dual of a free spectrahedrop is again a free spectrahedrop (Theorem 4.11). Section 5 contains the "perfect" Convex Positivstellensatz for free polynomials positive semidefinite on free spectrahedrops. The proof depends upon the results of Section 4. In Section 6 we introduce tracial sets and hulls and discuss their connections with the quantum interpolation problems from Section 3. Finally, Section 7 introduces tracial spectrahedra and proves a Hahn–Banach separation theorem in the tracial context (Theorem 7.6). This theorem is then used to suggest corresponding notions of duality. Section 8 contains examples.

## 2. Preliminaries

This section introduces terminology and presents preliminaries on free polynomials, free sets and free convexity that we need.

# 2.1. Free sets

A set  $\Gamma \subseteq \mathbb{S}^g$  is closed with respect to (*simultaneous*) *unitary conjugation* if for each n, each  $A \in \Gamma(n)$  and each  $n \times n$  unitary matrix U,

$$U^*AU = (U^*A_1U, \dots, U^*A_gU) \in \Gamma(n).$$

The set  $\Gamma$  is a *free set* if it is closed with respect to direct sums (see (1.1)) and simultaneous unitary conjugation. In particular, a matrix convex set is a free set. We refer the reader to [Voi04, Voi10, KVV14, MS11, Pop10, AM15, BB07] for a systematic study of free sets and free function theory. The set  $\Gamma$  is (*uniformly*) *bounded* if there is a  $C \in \mathbb{R}_{>0}$  such that  $C - \sum X_i^2 \succeq 0$  for all  $X \in \Gamma$ .

### 2.2. Free polynomials

One natural way free sets arise is as the nonnegativity set of a free polynomial. Given positive integers  $\ell$  and  $\nu$ , let  $\mathbb{C}^{\ell \times \nu}$  denote the collection of  $\ell \times \nu$  complex matrices. An expression of the form

$$P = \sum_{w} B_{w} w \in \mathbb{C}^{\ell \times \nu} \langle x \rangle,$$

where  $B_w \in \mathbb{C}^{\ell \times \nu}$ , and the sum is a finite sum over the words in the variables *x*, is a *free (noncommutative) matrix-valued polynomial*. The collection of all  $\ell \times \nu$ -valued free polynomials is denoted  $\mathbb{C}^{\ell \times \nu} \langle x \rangle$ , and  $\mathbb{C} \langle x \rangle$  denotes the set of scalar-valued free polynomials. We use  $\mathbb{C}^{\ell \times \nu} \langle x \rangle_k$  to denote free polynomials of degree  $\leq k$ . Here the *degree* of a word is its length. The free polynomial *P* is evaluated at an  $X \in \mathbb{S}_n^g$  by

$$P(X) = \sum_{w \in \langle x \rangle} B_w \otimes w(X) \in \mathbb{C}^{\ell n \times \mu n}$$

where  $\otimes$  denotes the (Kronecker) tensor product.

There is a natural involution \* on words that reverses the order. This involution extends to  $\mathbb{C}^{\ell \times \nu} \langle x \rangle$  by

$$P^* = \sum_w B^*_w w^* \in \mathbb{C}^{\mu \times \ell} \langle x \rangle.$$

If  $\mu = \ell$  and  $P^* = P$ , then *P* is *symmetric*. Note that if  $P \in \mathbb{C}^{\ell \times \ell} \langle x \rangle$  is symmetric, and  $X \in \mathbb{S}_n^g$ , then  $P(X) \in \mathbb{C}^{\ell n \times \ell n}$  is a hermitian matrix.

#### 2.3. Free semialgebraic sets

The *nonnegativity set*  $\mathcal{D}_P \subseteq \mathbb{S}^g$  of a symmetric free polynomial is the sequence of sets

$$\mathcal{D}_P(n) = \{ X \in \mathbb{S}_n^g : P(X) \succeq 0 \}.$$

It is readily checked that  $\mathcal{D}_P$  is a free set. By analogy with (commutative) real algebraic geometry, we call  $\mathcal{D}_P$  a basic *free semialgebraic set*. Often it is assumed that P(0) > 0. The free set  $\mathcal{D}_P$  has the additional property that it is *closed with respect to restriction to reducing subspaces*, that is, if  $X \in \mathcal{D}_P(n)$  and  $\mathcal{H} \subseteq \mathbb{C}^n$  is a reducing (equivalently invariant) subspace for X of dimension m, then X restricted to  $\mathcal{H}$  is in  $\mathcal{D}_P(m)$ .

## 2.4. Free convexity

If  $\Gamma$  is matrix convex, it is easy to show that  $\Gamma$  is levelwise convex. More generally, if  $A^{\ell} = (A_1^{\ell}, \ldots, A_g^{\ell})$  are in  $\Gamma(n_{\ell})$  for  $1 \leq \ell \leq k$ , then  $A = \bigoplus_{\ell=1}^k A^{\ell} \in \Gamma(n)$ , where  $n = \sum n_{\ell}$ . Hence, if  $V_{\ell}$  are  $n_{\ell} \times m$  matrices (for some *m*) such that  $V = (V_1^* \ldots V_k^*)^*$  is an isometry (equivalently  $\sum_{\ell=1}^k V_{\ell}^* V_{\ell} = I_m$ ), then

$$V^*AV = \sum_{\ell=1}^{k} V_{\ell}^* A^{\ell} V_{\ell} \in \Gamma(m).$$
 (2.1)

A sum as in (2.1) is a matrix (free) convex combination of the g-tuples  $\{A^{\ell} : \ell = 1, \ldots, k\}$ .

**Lemma 2.1** ([HKM16, Lemma 2.3]). Suppose  $\Gamma$  is a free subset of  $\mathbb{S}^g$ .

- (1) If  $\Gamma$  is closed with respect to restriction to reducing subspaces, then the following are equivalent:
  - (i)  $\Gamma$  is matrix convex;
  - (ii)  $\Gamma$  is levelwise convex.
- (2) If  $\Gamma$  is (nonempty and) matrix convex, then  $0 \in \Gamma(1)$  if and only if  $\Gamma$  is closed with respect to (*simultaneous*) conjugation by contractions.

Convex subsets of  $\mathbb{R}^g$  are defined as intersections of half-spaces and are thus described by linear functionals. Analogously, matrix convex subsets of  $\mathbb{S}^g$  are defined by linear pencils [EW97, HM12]. We next present basic facts about linear pencils and their associated matrix convex sets.

2.4.1. Linear pencils. Recall the definition (see (1.2)) of the (affine) linear pencil  $L_A(x)$  associated to a tuple  $A = (A_0, \ldots, A_g) \in \mathbb{S}_k^{g+1}$ . If  $A_0 = 0$ , i.e.,  $A = (A_1, \ldots, A_g) \in \mathbb{S}_k^g$ , let

$$\Lambda_A(x) = \sum_{j=1}^g A_j x_j$$

denote the corresponding homogeneous (truly) linear pencil, and

$$\mathfrak{L}_A = I - \Lambda_A$$

the associated monic linear pencil.

The pencil  $L_A$  (and also  $\mathfrak{L}_A$ ) is a free polynomial with matrix coefficients, so it is naturally evaluated on  $X \in \mathbb{S}_n^g$  using (Kronecker's) tensor product, yielding (1.3). The free semialgebraic set  $\mathcal{D}_{L_A}$  is easily seen to be matrix convex. We will refer to  $\mathcal{D}_{L_A}$  as a *free spectrahedron* or *free LMI domain*, and say that a free set  $\Gamma$  is *freely LMI representable* if there is a linear pencil L such that  $\Gamma = \mathcal{D}_L$ . In particular, if  $\Gamma$  is freely LMI representable with a monic  $\mathfrak{L}_A$ , then 0 is in the interior of  $\Gamma(1)$ .

The following is a special case of a theorem due to Effros and Winkler [EW97] (see also [HKM13a, Theorem 3.1]). Given a free set  $\Gamma$ , if  $0 \in \Gamma(1)$ , then  $0 \in \Gamma(n)$  for each *n*. In this case we will write  $0 \in \Gamma$ .

**Theorem 2.2.** If  $C = (C(n))_{n \in \mathbb{N}} \subseteq \mathbb{S}^g$  is a closed matrix convex set containing 0 and  $Y \in \mathbb{S}^g_m$  is not in C(m), then there is a monic linear pencil  $\mathfrak{L}$  of size m such that  $\mathfrak{L}(X) \succeq 0$  for all  $X \in C$ , but  $\mathfrak{L}(Y) \succeq 0$ .

By the following result from [HM12], linear matrix inequalities account for matrix convexity of free semialgebraic sets.

**Theorem 2.3.** Fix a symmetric matrix polynomial p. If p(0) > 0 and the strict positivity set  $\mathfrak{P}_p = \{X : p(X) > 0\}$  of p is bounded, then  $\mathfrak{P}_p$  is matrix convex if and only if it is freely LMI representable with a monic pencil.

## 2.5. A Convex Positivstellensatz and LMI domination

Positivstellensätze are pillars of real algebraic geometry [BCR98]. We next recall the Positivstellensätz for a free polynomial p. It is the algebraic certificate for nonnegativity of pon the free spectrahedron  $\mathcal{D}_L$  from [HKM12]. It is "perfect" in the sense that p is only assumed to be nonnegative on  $\mathcal{D}_L$ , and we obtain degree bounds on the scale of deg(p)/2for the polynomials involved in the positivity certificate. In Section 5, we will extend this Positivstellensatz to free spectrahedrops (i.e., projections of free spectrahedra)—see Theorem 5.1.

**Theorem 2.4.** Suppose  $\mathfrak{L}$  is a monic linear pencil. A matrix polynomial p is positive semidefinite on  $\mathcal{D}_{\mathfrak{L}}$  if and only if it has a weighted sum of squares representation with optimal degree bounds:

$$p = s^* s + \sum_j^{\text{finite}} f_j^* \mathfrak{L} f_j,$$

where s,  $f_i$  are matrix polynomials of degree no greater than deg(p)/2.

In particular, if  $\mathfrak{L}_A$ ,  $\mathfrak{L}_B$  are monic linear pencils, then  $\mathcal{D}_{\mathfrak{L}_B} \subseteq \mathcal{D}_{\mathfrak{L}_A}$  if and only if there exists a positive integer  $\mu$  and a contraction V such that

$$A = V^* (I_\mu \otimes B) V. \tag{2.2}$$

If  $\mathcal{D}_{\mathfrak{L}_B}$  is bounded, then V can be chosen to be an isometry.

*Proof.* The first statement is [HKM12, Theorem 1.1]. Applying this result to the *LMI* domination problem  $\mathcal{D}_{\mathfrak{L}_B} \subseteq \mathcal{D}_{\mathfrak{L}_A}$ , we see that  $\mathcal{D}_{\mathfrak{L}_B} \subseteq \mathcal{D}_{\mathfrak{L}_A}$  is equivalent to

$$\mathfrak{L}_A(x) = S^*S + \sum_{j=1}^{\mu} V_j^* \mathfrak{L}_B(x) V_j$$
(2.3)

for some matrices  $S, V_i$ , i.e.,

$$I = S^*S + \sum_{j} V_j^* V_j = S^*S + V^*V, \qquad (2.4)$$

$$A = \sum_{j=1}^{\mu} V_j^* B V_j = V^* (I_{\mu} \otimes B) V, \qquad (2.5)$$

where V is the block column matrix of the  $V_j$ . Equation (2.4) simply says that V is a contraction, and (2.5) is (2.2). The last statement is proved in [HKM13a]. Alternatively, as is shown in [HKM12, Proposition 4.2], if  $\mathcal{D}_{\mathfrak{L}_B}$  is bounded, then there are finitely many matrices  $W_j$  such that

$$= \sum W_j^* \mathfrak{L}_B(x) W_j.$$

I

Writing  $S^*S = \sum (W_j S)^* \mathfrak{L}_B(x)(W_j S)$  and inserting into (2.3) completes the proof.  $\Box$ 

Example 8.1 shows that it is not necessarily possible to choose V in (2.5) to be an isometry without additional hypothesis on the tuple B.

## 3. Completely positive interpolation

Theorem 3.4 below provides a solution to three cp interpolation problems in terms of concrete LMIs that can be solved with a standard semidefinite programming (SDP) solver. The unital cp interpolation problem comes from efforts to understand matrix convex sets that arise in convex optimization. Its solution plays an important role in the proof of the main result on the polar dual of a free spectrahedrop (Theorem 4.11) via its appearance in the proof of Proposition 4.14.

The trace preserving and trace nonincreasing cp interpolation problems arise in quantum information theory and the study of quantum channels, where one is interested in sending one (finite) set of prescribed quantum states to another.

#### 3.1. Basics of completely positive maps

This subsection collects basic facts about completely positive (cp) maps  $\phi : S \to M_d$ , where S is a subspace of  $M_n$  closed under conjugate transpose (see for instance [Pau02]) containing a positive definite matrix. Thus S is an *operator system*.

Suppose S is a subspace of  $M_n$  closed under conjugate transpose,  $\phi : S \to M_d$  is a linear map and  $\ell$  is a positive integer. The ( $\ell$ -th) *ampliation*  $\phi_{\ell} : M_{\ell}(S) \to M_{\ell}(M_d)$  of  $\phi$  is defined by applying  $\phi$  entrywise,

$$\phi_{\ell}(S_{j,k}) = \left(\phi(S_{j,k})\right).$$

The map  $\phi$  is *symmetric* if  $\phi(S^*) = \phi(S)^*$ , and it is *completely positive* if each  $\phi_\ell$  is positive in the sense that if  $S \in M_\ell(S)$  is positive semidefinite, then so is  $\phi_\ell(S) \in M_\ell(M_d)$ . In what follows, S is often a subspace of  $\mathbb{S}_n$  (and is thus automatically closed under the conjugate transpose operation).

The *Choi matrix* of a mapping  $\phi : M_n \to M_d$  is the  $n \times n$  block matrix with  $d \times d$  matrix entries given by

$$(C_{\phi})_{i,j} = (\phi(E_{i,j}))_{i,j}.$$

On the other hand, a matrix  $C = (C_{i,j}) \in M_n(M_d)$  determines a mapping  $\phi_C : M_n \to M_d$ by  $\phi_C(E_{i,j}) = C_{i,j} \in M_d$ . A matrix *C* is a *Choi matrix for*  $\phi : S \to M_d$  if the mapping  $\phi_C$  agrees with  $\phi$  on *S*. **Theorem 3.1.** For  $\phi : M_n \to M_d$ , the following are equivalent:

- (a)  $\phi$  is completely positive;
- (b) the Choi matrix  $C_{\phi}$  is positive semidefinite.

Suppose  $S \subseteq M_n$  is an operator system. For a symmetric  $\phi : S \to M_d$ , the following are equivalent:

- (i)  $\phi$  is completely positive;
- (ii)  $\phi_d$  is positive;
- (iii) there exists a completely positive mapping  $\Phi: M_n \to M_d$  extending  $\phi$ ;
- (iv) there is a positive semidefinite Choi matrix for  $\phi$ ;
- (v) there exist  $n \times d$  matrices  $V_1, \ldots, V_{nd}$  such that

$$\phi(A) = \sum V_j^* A V_j. \tag{3.1}$$

Finally, for a subspace S of  $M_n$ , a mapping  $\phi : S \to M_d$  has a completely positive extension  $\Phi : M_n \to M_d$  if and only if  $\phi$  has a positive semidefinite Choi matrix.

**Lemma 3.2.** The cp mapping  $\phi : M_n \to M_d$  as in (3.1) is

(a) unital (that is,  $\phi(I_n) = I_d$ ) if and only if

$$\sum_{j} V_j^* V_j = I;$$

(b) trace preserving if and only if

$$\sum_{j} V_j V_j^* = I;$$

(c) trace nonincreasing for positive semidefinite matrices (i.e.,  $tr(\phi(P)) \le tr(P)$  for all positive semidefinite P) if and only if

$$\sum_j V_j V_j^* \preceq I.$$

*Proof.* We prove (c), and leave (a) and (b) as an easy exercise for the reader. For  $A \in M_n$ ,

$$\operatorname{tr}(\phi(A)) = \sum_{j} \operatorname{tr}(V_{j}^{*}AV_{j}) = \operatorname{tr}\left(A\sum_{j}V_{j}V_{j}^{*}\right).$$

Hence the trace nonincreasing property for  $\phi$  is equivalent to

$$\operatorname{tr}\left(P\left(I-\sum_{j}V_{j}V_{j}^{*}\right)\right)\geq0$$

for all positive semidefinite P, i.e.,  $I - \sum_{j} V_{j}V_{j}^{*} \succeq 0$ .

# **Proposition 3.3.** The linear mapping $\phi : M_n \to M_d$ is

(a) unital (that is,  $\phi(I_n) = I_d$ ) if and only if its Choi matrix C satisfies

$$\sum_{j=1}^{n} C_{j,j} = I;$$

(b) trace preserving if and only if its Choi matrix C satisfies

$$(tr(C_{i,j}))_{i,j=1}^n = I_n;$$

(c) trace nonincreasing for positive semidefinite matrices (i.e.,  $tr(\phi(P)) \le tr(P)$  for all positive semidefinite P) if and only if its Choi matrix C satisfies

$$(\operatorname{tr}(C_{i,j}))_{i,j} \leq I_n,$$

Proof. Statement (a) follows from

$$\phi(I_n) = \phi\left(\sum_{j=1}^n E_{j,j}\right) = \sum_{j=1}^n C_{j,j},$$

where C is the Choi matrix for  $\phi$ . Here  $E_{i,j}$  denote the matrix units.

For (b), let  $X = \sum_{i,j=1}^{n} \alpha_{i,j} E_{i,j}$ . Then

$$\operatorname{tr}(X) = \sum_{i=1}^{n} \alpha_{i,i}, \quad \operatorname{tr}(\phi(X)) = \sum_{i,j=1}^{n} \alpha_{i,j} \operatorname{tr}(C_{i,j}).$$

Since  $tr(\phi(X)) = tr(X)$  for all X, this linear system yields  $tr(C_{i,j}) = \delta_{i,j}$  for all i, j.

Finally, for (c), if  $\phi$  is trace nonincreasing, choosing  $X = xx^*$  a rank one matrix,  $X = (x_i x_j)$ , we find that

$$\sum x_i x_j \operatorname{tr}(C_{i,j}) = \operatorname{tr}(\phi(X)) \le \operatorname{tr}(X) = \sum x_i^2.$$

Hence  $I - (tr(C_{i,j})) \ge 0$ . Conversely, if  $I - (tr(C_{i,j})) \ge 0$ , then for any positive semidefinite rank one matrix X, the computation above shows that  $tr(\phi(X)) \le tr(X)$ . Finally, use the fact that any positive semidefinite matrix is a sum of rank one positive semidefinite matrices to complete the proof.

The Arveson extension theorem [Arv69] says that any cp (resp. ucp) map on an operator system extends to a cp (resp. ucp) map on the full algebra. Example 8.2 shows that a CPTP map need not extend to a CPTP map on the full algebra.

## 3.2. Quantum interpolation problems and semidefinite programming

The cp interpolation problem is formulated as follows. Given  $A^1 \in \mathbb{S}_n^g$  and  $A^2 \in \mathbb{S}_m^g$ , does there exist a cp map  $\Phi : M_n \to M_m$  such that

$$A_{\ell}^{2} = \Phi(A_{\ell}^{1})$$
 for  $\ell = 1, ..., g$ ?

One can require further that

- (1)  $\Phi$  be *unital*, or
- (2)  $\Phi$  be *trace preserving*, or
- (3) Φ be *trace nonincreasing* in the sense that tr(Φ(P)) ≤ tr(P) for all positive semidefinite P.

Our solutions to these interpolation problems are formulated as concrete LMIs that can be solved with a standard semidefinite programming (SDP) solver. They are equivalent to, but stated quite differently than, the earlier results in [AG15].

**Theorem 3.4.** Suppose for  $\ell = 1, ..., g$  the matrices  $A_{\ell}^1 \in \mathbb{S}_n$  and  $A_{\ell}^2 \in \mathbb{S}_m$  are given. Let  $\alpha_{p,q}^{\ell}$  denote the (p,q)-entry of  $A_{\ell}^1$ . There exists a cp map  $\Phi : M_n \to M_m$  that solves the interpolation problem

$$\Phi(A^1_\ell) = A^2_\ell, \quad \ell = 1, \dots, g,$$

if and only if the following feasibility semidefinite programming problem has a solution:

$$(C_{p,q})_{p,q=1}^{n} =: C \succeq 0, \quad \forall \ell = 1, \dots, g, \ \sum_{p,q}^{n} \alpha_{p,q}^{\ell} C_{p,q} = A_{\ell}^{2},$$
(3.2)

for the unknown  $mn \times mn$  symmetric matrix  $C = (C_{p,q})_{p,q=1}^n$  consisting of  $m \times m$  blocks  $C_{p,q}$ . Furthermore,

(1) the map  $\Phi$  is unital if and only if, in addition to (3.2),

$$\sum_{p=1}^{n} C_{p,p} = I_m;$$
(3.3)

(2) the map  $\Phi$  is a quantum channel if and only if, in addition to (3.2),

$$(\operatorname{tr}(C_{p,q}))_{p,q} = I_n; \tag{3.4}$$

(3) the map  $\Phi$  is a quantum operation if and only if, in addition to (3.2),

$$(\operatorname{tr}(C_{p,q}))_{p,q} \leq I_n. \tag{3.5}$$

In each case the constraints on C are LMIs, and the set of solutions C constitutes a bounded spectrahedron.

**Remark 3.5.** In the unital case the resulting spectrahedron is free: for fixed  $A^1 \in \mathbb{S}_n^g$ , the sequence of solution sets to (3.2) and (3.3) parametrized over *m* is a free spectrahedron (see Proposition 4.14 for details). In the two quantum cases, for each *m*, the solutions  $\mathcal{D}(m)$  at level *m* form a spectrahedron, but the sequence  $\mathcal{D} = (\mathcal{D}(m))_m$  is in general not a free spectrahedron since it fails to respect direct sums.

*Proof of Theorem 3.4.* This interpolation result is a consequence of Theorem 3.1. Let S denote the span of  $\{A_{\ell}^1\}$  and  $\phi$  the mapping from S to  $M_m$  defined by  $\phi(A_{\ell}^1) = A_{\ell}^2$ . This mapping has a completely positive extension  $\Phi : M_n \to M_m$  if and only if it has a positive semidefinite Choi matrix. The conditions on C evidently are exactly those needed to say that C is a positive semidefinite Choi matrix for  $\phi$ .

The additional conditions in (3.3) and (3.4) (i.e.,  $\phi(I_n) = I_m$  and trace preservation) are clearly linear, so produce a spectrahedron in  $\mathbb{S}_{mn}$ . Both spectrahedra are bounded. Indeed, in each case  $C_{p,p} \leq I_m$ , so  $C \leq I_{mn}$ . Likewise, the additional condition in (3.5) is an LMI constraint, producing a bounded spectrahedron.

We note that cp maps between subspaces of matrix algebras in the absence of positive definite elements were treated in [HKN14, Section 8] (see also [KS13, KTT13]).

## 4. Free spectrahedrops and polar duals

This section starts by recalling the definition of a free spectrahedrop as the coordinate projection of a spectrahedron. It then continues with a review of free polar duals [EW97] and their basic properties, before turning to two main results, stated now without technical hypotheses. Firstly, a free convex set is, in a canonical sense, generated by a finite set (equivalently a single point) if and only if it is the polar dual of a free spectrahedron (Theorem 4.6). Secondly, the polar dual of a free spectrahedrop is again a free spectrahedrop (Theorem 4.11).

## 4.1. Projections of free spectrahedra: free spectrahedrops

Let *L* be a linear pencil in the variables  $(x_1, \ldots, x_g; y_1, \ldots, y_h)$ . Thus, for some *d* and  $d \times d$  hermitian matrices  $D, \Omega_1, \ldots, \Omega_g, \Gamma_1, \ldots, \Gamma_h$ ,

$$L(x, y) = D + \sum_{j=1}^{g} \Omega_j x_j + \sum_{\ell=1}^{h} \Gamma_\ell y_\ell.$$

The set

$$\operatorname{proj}_{x} \mathcal{D}_{L}(1) = \{ x \in \mathbb{R}^{g} : \exists y \in \mathbb{R}^{h}, \ L(x, y) \succeq 0 \}$$

is known as a *spectrahedral shadow* or a *semidefinite programming (SDP) representable set* [BPR13] and the representation afforded by *L* is an *SDP representation*. SDP representable sets are evidently convex and lie in a middle ground between LMI representable sets and general convex sets. They play an important role in convex optimization [Nem06]. If  $S \subseteq \mathbb{R}^g$  is closed semialgebraic and satisfies some mild additional hypothesis, it is proved in [HN10] based upon the Lasserre–Parrilo construction [Las09, Par06] that the convex hull of *S* is SDP representable. Given a linear pencil L, let  $\operatorname{proj}_{X} \mathcal{D}_{L} = (\operatorname{proj}_{X} \mathcal{D}_{L}(n))_{n}$  denote the free set

$$\operatorname{proj}_{x} \mathcal{D}_{L}(n) = \{ X \in \mathbb{S}_{n}^{g} : \exists Y \in \mathbb{S}_{n}^{h}, \ L(X, Y) \succeq 0 \}.$$

We call a set of the form  $\operatorname{proj}_x \mathcal{D}_L$  a *free spectrahedrop* and  $\mathcal{D}_L$  an *LMI lift* of  $\operatorname{proj}_x \mathcal{D}_L$ . Thus a free spectrahedrop is a coordinate projection of a free spectrahedron. Clearly, free spectrahedrops are matrix convex. In particular, they are closed with respect to restrictions to reducing subspaces.

**Lemma 4.1** ([HKM16, §4.1]). If  $\mathcal{K} = \operatorname{proj}_{x} \mathcal{D}_{L}$  is a free spectrahedrop containing  $0 \in \mathbb{R}^{g}$  in the interior of  $\mathcal{K}(1)$ , then there exists a monic linear pencil  $\mathfrak{L}(x, y)$  such that

$$\mathcal{K} = \operatorname{proj}_{X} \mathcal{D}_{\mathfrak{L}} = \{ X \in \mathbb{S}^{g} : \exists Y \in \mathbb{S}^{h}, \ \mathfrak{L}(X, Y) \succeq 0 \}.$$

If, in addition,  $\mathcal{D}_L$  is bounded, then we may further ensure  $\mathcal{D}_{\mathfrak{L}}$  is bounded.

If the free spectrahedrop  $\mathcal{K}$  is closed and bounded, and contains 0 in its interior, then there is a monic linear pencil  $\mathfrak{L}$  such that  $\mathcal{D}_{\mathfrak{L}}$  is bounded and  $\mathcal{K} = \operatorname{proj}_{\mathfrak{X}} \mathcal{D}_{\mathfrak{L}}$  (see Theorem 4.11).

Let  $p = 1 - x_1^2 - x_2^4$ . It is well known that  $\mathcal{D}_p(1) = \{(x, y) \in \mathbb{R}^2 : 1 - x_1^2 - x_2^4 \ge 0\}$  is a spectrahedral shadow. On the other hand,  $\mathcal{D}_p(2)$  is not convex (in the usual sense), and hence  $\mathcal{D}_p$  is not a spectrahedrop. Further details can be found in Example 8.3.

#### 4.2. Basics of polar duals

By precise analogy with the classical  $\mathbb{R}^g$  notion, the *free polar dual*  $\mathcal{K}^\circ = (\mathcal{K}^\circ(n))_n$  of a free set  $\mathcal{K} \subseteq \mathbb{S}^g$  is

$$\mathcal{K}^{\circ}(n) := \left\{ A \in \mathbb{S}_{n}^{g} : \mathfrak{L}_{A}(X) = I \otimes I - \sum_{j=1}^{g} A_{j} \otimes X_{j} \succeq 0 \text{ for all } X \in \mathcal{K} \right\}$$

Given  $\varepsilon > 0$ , consider the free  $\varepsilon$ -ball centered at 0,

$$\mathcal{N}_{\varepsilon} := \{ X \in \mathbb{S}^g : \|X\| \le \varepsilon \} = \left\{ X : \varepsilon^2 I \succeq \sum_j X_j^2 \right\}.$$

It is easy to see that its polar dual is bounded. In fact,

$$\mathcal{N}_{1/(g\varepsilon)} \subseteq \mathcal{N}_{\varepsilon}^{\circ} \subseteq \mathcal{N}_{\sqrt{g}/\varepsilon}$$

We say that 0 *is in the interior* of the subset  $\Gamma \subseteq \mathbb{S}^g$  if  $\Gamma$  contains some free  $\varepsilon$ -ball centered at 0.

**Lemma 4.2.** Suppose  $\mathcal{K} \subseteq \mathbb{S}^g$  is matrix convex. The following are equivalent:

- (i)  $0 \in \mathbb{R}^{g}$  is in the interior of  $\mathcal{K}(1)$ ;
- (ii)  $0 \in \mathbb{S}_n^g$  is in the interior of  $\mathcal{K}(n)$  for some n;
- (iii)  $0 \in \mathbb{S}_n^g$  is in the interior of  $\mathcal{K}(n)$  for all n;
- (iv) 0 is in the interior of  $\mathcal{K}$ .

*Proof.* It is clear that (iv) $\Rightarrow$ (iii) $\Rightarrow$ (ii). Assume (ii) holds. There is an  $\varepsilon > 0$  with  $\mathcal{N}_{\varepsilon}(n) \subseteq \mathcal{K}(n)$ . Since  $\mathcal{K}$  is closed with respect to restriction to reducing subspaces, and

$$\mathcal{N}_{\varepsilon}(1) \oplus \cdots \oplus \mathcal{N}_{\varepsilon}(1) \subseteq \mathcal{N}_{\varepsilon}(n),$$

we see  $\mathcal{N}_{\varepsilon}(1) \subseteq \mathcal{K}(1)$ , i.e., (i) holds.

Now suppose (i) holds, i.e.,  $\mathcal{N}_{\varepsilon}(1) \subseteq \mathcal{K}(1)$  for some  $\varepsilon > 0$ . We claim that  $\mathcal{N}_{\varepsilon/g^2} \subseteq \mathcal{K}$ . Let  $X \in \mathcal{N}_{\varepsilon/g^2}$ . It is clear that  $[-\varepsilon/g, \varepsilon/g]^g \subseteq \mathcal{K}(1)$ , hence  $[-\varepsilon/g, \varepsilon/g]^g \otimes I_n \subseteq \mathcal{K}(n)$ . Since each  $X_j$  has norm  $\leq \varepsilon/g^2$ , matrix convexity of  $\mathcal{K}$  implies that

$$(0,\ldots,0,gX_i,0,\ldots,0)\in\mathcal{K},$$

and thus

$$X = \frac{1}{g} ((gX_1, 0, \dots, 0) + \dots + (0, \dots, 0, gX_g)) \in \mathcal{K}.$$

For the readers' convenience, the following proposition lists some properties of  $\mathcal{K}^{\circ}$ . The bipolar result of item (6) is due to [EW97]. Given a collection  $\Gamma_{\alpha}$  of matrix convex sets, it is readily verified that  $\Gamma = (\Gamma(n))_n$  defined by  $\Gamma(n) = \bigcap_{\alpha} \Gamma_{\alpha}(n)$  is again matrix convex. Likewise, if  $\Gamma$  is matrix convex, then so is its closure  $\overline{\Gamma} = (\overline{\Gamma(n)})_n$ . Given a subset  $\mathcal{K}$  of  $\mathbb{S}^g$ , let  $\operatorname{co}^{\operatorname{mat}} \mathcal{K}$  denote the intersection of all matrix convex sets containing  $\mathcal{K}$ . Thus,  $\operatorname{co}^{\operatorname{mat}} \mathcal{K}$  is the smallest matrix convex set containing  $\mathcal{K}$ . Likewise,  $\overline{\operatorname{co}^{\operatorname{mat}} \mathcal{K} = \overline{\operatorname{co}^{\operatorname{mat}} \mathcal{K}}$  is the smallest closed matrix convex set containing  $\mathcal{K}$ . Details, and an alternative characterization of the matrix convex hull of a free set  $\mathcal{K}$ , can be found in [HKM16].

#### **Proposition 4.3.** Suppose $\mathcal{K} \subseteq \mathbb{S}^{g}$ .

(1)  $\mathcal{K}^{\circ}$  is a closed matrix convex set containing 0;

(2) if 0 is in the interior of  $\mathcal{K}$ , then  $\mathcal{K}^{\circ}$  is bounded;

(3)  $\mathcal{K}(n) \subseteq \mathcal{K}^{\circ\circ}(n)$  for all n, that is,  $\mathcal{K} \subseteq \mathcal{K}^{\circ\circ}$ ;

(4)  $\mathcal{K}$  is bounded if and only if 0 is in the interior of  $\mathcal{K}^{\circ}$ ;

(5) *if there is an m such that*  $0 \in \mathcal{K}(m)$ *, then*  $\mathcal{K}^{\circ\circ} = \overline{\operatorname{co}}^{\mathrm{mat}} \mathcal{K}$ *;* 

(6) if  $\mathcal{K}$  is a closed matrix convex set containing 0, then  $\mathcal{K} = \mathcal{K}^{\circ\circ}$ ;

(7) if  $\mathcal{K}$  is matrix convex, then  $\mathcal{K}(1)^{\circ} = \mathcal{K}^{\circ}(1)$ .

*Proof.* Matrix convexity in (1) is straightforward.

If  $\mathcal{K}$  has 0 in its interior, then there is a small free neighborhood  $\mathcal{N}_{\varepsilon}$  of 0 inside  $\mathcal{K}$ . Hence  $\mathcal{K}^{\circ} \subseteq \mathcal{N}_{\varepsilon}^{\circ} = \mathcal{N}_{1/\varepsilon}$  is bounded.

Item (3) is a tautology. Indeed, if  $X \in \mathcal{K}(n)$ , then we want to show  $\mathfrak{L}_X(A) \succeq 0$ whenever  $\mathfrak{L}_A(Y) \succeq 0$  for all Y in  $\mathcal{K}$ . But this follows simply from the fact that  $\mathfrak{L}_X(A)$ and  $\mathfrak{L}_A(X)$  are unitarily equivalent.

If  $\mathcal{K}$  is bounded, then it is evident that 0 is in the interior of  $\mathcal{K}^{\circ}$ . If 0 is in the interior of  $\mathcal{K}^{\circ}$ , then, by (2),  $\mathcal{K}^{\circ\circ}$  is bounded. By (3),  $\mathcal{K} \subseteq \mathcal{K}^{\circ\circ}$ , and thus  $\mathcal{K}$  is bounded.

To prove (5), first note that  $0 \in \overline{\operatorname{co}}^{\operatorname{mat}} \mathcal{K}(m)$  and since  $\overline{\operatorname{co}}^{\operatorname{mat}} \mathcal{K}(m)$  is matrix convex,  $0 \in \overline{\operatorname{co}}^{\operatorname{mat}} \mathcal{K}(1)$ . Now suppose  $W \notin \overline{\operatorname{co}}^{\operatorname{mat}} \mathcal{K}$ . The Effros–Winkler matricial Hahn– Banach Theorem 2.2 produces a monic linear pencil  $\mathfrak{L}_A$  (with the size of A no larger than the size of W) separating W from  $\overline{\operatorname{co}}^{\operatorname{mat}} \mathcal{K}$ , that is,  $\mathfrak{L}_A(W) \succeq 0$  and  $\mathfrak{L}_A(X) \succeq 0$  for  $X \in \operatorname{co}^{\operatorname{mat}} \mathcal{K}$ . Hence  $A \in \mathcal{K}^{\circ}$ . Using the unitary equivalence of  $\mathfrak{L}_W(A)$  and  $\mathfrak{L}_A(W)$  we see that  $\mathfrak{L}_W(A) \not\geq 0$ , and thus  $W \notin \mathcal{K}^{\circ\circ}$ . Thus,  $\mathcal{K}^{\circ\circ} \subseteq \overline{\operatorname{co}}^{\operatorname{mat}} \mathcal{K}$ . The reverse inclusion follows from (3).

Finally, suppose  $\mathcal{K}$  is matrix convex and  $y \in \mathcal{K}(1)^\circ$ . Thus,  $\sum y_j x_j = \langle y, x \rangle \le 1$  for all  $x \in \mathcal{K}(1)$ . Given  $X \in \mathcal{K}(m)$  and a unit vector  $v \in \mathbb{C}^m$ , since  $v^* X v \in \mathcal{K}(1)$ , we have

$$1 \ge \sum y_j v^* X_j v.$$

Hence,

$$v^* \Big( I - \sum y_j X_j \Big) v \ge 0$$

for all unit vectors v. So  $y \in \mathcal{K}^{\circ}(1)$ . The reverse inclusion is immediate.

**Corollary 4.4.** If  $\mathcal{K} \subseteq \mathbb{S}^g$ , then  $\mathcal{K}^{\circ\circ} = \overline{\operatorname{co}}^{\operatorname{mat}}(\mathcal{K} \cup \{0\})$ . Here  $0 \in \mathbb{R}^g$ .

*Proof.* Note that  $\mathcal{K}^{\circ} = (\mathcal{K} \cup \{0\})^{\circ}$ , and hence  $\mathcal{K}^{\circ\circ} = (\mathcal{K} \cup \{0\})^{\circ\circ}$ . By Proposition 4.3(5),

$$\overline{\operatorname{co}}^{\operatorname{mat}}(\mathcal{K} \cup \{0\}) = (\mathcal{K} \cup \{0\})^{\circ \circ}.$$

**Lemma 4.5.** Suppose  $\mathcal{K} \subseteq \mathbb{S}^{g+h}$ , and consider its image proj  $\mathcal{K} \subseteq \mathbb{S}^{g}$  under the projection proj :  $\mathbb{S}^{g+h} \to \mathbb{S}^{g}$ . A tuple  $A \in \mathbb{S}^{g}$  is in (proj  $\mathcal{K}$ )° if and only if  $(A, 0) \in \mathcal{K}^{\circ}$ .

*Proof.* Note that  $A \in (\operatorname{proj} \mathcal{K})^{\circ}$  if and only if for all  $X \in \operatorname{proj} \mathcal{K}$  we have  $\mathfrak{L}_A(X) \succeq 0$ , if and only if  $\mathfrak{L}_{(A,0)}(X,Y) \succeq 0$  for all  $X \in \operatorname{proj} \mathcal{K}$  and all  $Y \in \mathbb{S}^h$ , if and only if  $\mathfrak{L}_{(A,0)}(X,Y) \succeq 0$  for all  $(X,Y) \in \mathcal{K}$ , if and only if  $(A,0) \in \mathcal{K}^{\circ}$ .  $\Box$ 

The polar dual of the set  $\{(x_1, x_2) \in \mathbb{R}^2 : 1 - x_1^2 - x_2^4 \ge 0\}$  is computed and seen not to be a spectrahedron in Example 8.4.

#### 4.3. Polar duals of free spectrahedra

The next theorem completely characterizes finitely generated matrix convex sets  $\mathcal{K}$  containing 0 in their interior. Namely, such sets are exactly polar duals of bounded free spectrahedra.

**Theorem 4.6.** Suppose  $\mathcal{K}$  is a closed matrix convex set with 0 in its interior. If there is an  $\Omega \in \mathcal{K}$  such that for each  $X \in \mathcal{K}$  there is a  $\mu \in \mathbb{N}$  and an isometry V such that

$$X_j = V^*(I_\mu \otimes \Omega_j)V, \tag{4.1}$$

then

$$\mathcal{K}^{\circ} = \mathcal{D}_{\mathfrak{L}_{\Omega}},\tag{4.2}$$

where  $\mathfrak{L}_{\Omega}$  is the monic linear pencil  $\mathfrak{L}_{\Omega}(x) = I - \sum \Omega_i x_i$ .

Conversely, if there is an  $\Omega$  such that (4.2) holds, then  $\Omega \in \mathcal{K}$  and for each  $X \in \mathcal{K}$  there is an isometry V such that (4.1) holds.

A variant of Theorem 4.6 in which the condition that 0 is in the interior of  $\mathcal{K}$  is replaced by the weaker hypothesis that 0 is merely in  $\mathcal{K}$ , and of course with a slightly weaker conclusion, is separately stated in Proposition 4.9 below.

**Lemma 4.7.** Suppose  $\Omega \in \mathbb{S}_d^g$  and consider the monic linear pencil  $\mathfrak{L}_\Omega = I - \sum \Omega_j x_j$ .

(1) Let  $\Omega' = \Omega \oplus 0$  where  $0 \in \mathbb{S}_d^g$ . A tuple  $X \in \mathbb{S}^g$  is in  $\mathcal{D}_{\mathfrak{L}_\Omega}^\circ$  if and only if there is an isometry V such that

$$X_j = V^* (I \otimes \Omega'_j) V.$$

(2) If  $\mathcal{D}_{\mathfrak{L}_{\Omega}}$  is bounded, then  $X \in \mathbb{S}^g$  is in  $\mathcal{D}^{\circ}_{\mathfrak{L}_{\Omega}}$  if and only if there is an isometry V such that (4.1) holds.

**Remark 4.8.** As an alternative of (2),  $X \in \mathcal{D}^{\circ}_{\mathfrak{L}_{\Omega}}$  if and only if there exists a contraction *V* such that (4.1) holds.

*Proof of Lemma 4.7.* Note that  $X \in \mathcal{D}_{\mathfrak{L}_{\Omega}}^{\circ}$  if and only if  $\mathcal{D}_{\mathfrak{L}_{\Omega}} \subseteq \mathcal{D}_{\mathfrak{L}_{X}}$ . Thus if  $\mathcal{D}_{\mathfrak{L}_{\Omega}}$  is bounded, then the result follows directly from the last part of Theorem 2.4. On the other hand, if *X* has a representation as in (4.1), then evidently  $X \in \mathcal{D}_{\mathfrak{L}_{\Omega}}^{\circ}$ .

If  $\mathcal{D}_{\mathfrak{L}_{\Omega}}$  is not necessarily bounded and  $X \in \mathcal{D}^{\circ}_{\mathfrak{L}_{\Omega}}(m)$ , then, by Theorem 2.4,

$$X = \sum_{j=1}^{\mu} V_j^* \Omega V_j$$

for some  $\mu$  and operators  $V_i : \mathbb{C}^m \to \mathbb{C}^n$  such that

$$I - \sum V_j^* V_j \succeq 0.$$

There is a  $\nu > \mu$  and  $m \times n$  matrices  $V_{\mu+1}, \ldots, V_{\nu}$  such that

$$\sum_{j=1}^{\nu} V_j^* V_j = I.$$

Let

$$W_j = \begin{cases} \begin{pmatrix} V_j \\ 0 \end{pmatrix} & \text{for } 1 \le j \le \mu, \\ (0 \ V_j^*)^* & \text{for } \mu < j \le \nu. \end{cases}$$

With this choice of W, note that  $\sum W_i^* W_j = I_m$  and

$$\sum W_{j}^{*}\Omega_{j}^{\prime}W_{j} = \sum W_{j}^{*}(\Omega_{j} \oplus 0)W_{j} = \sum_{j=1}^{\nu} V_{j}^{*}\Omega_{j}V_{j} = X_{j}.$$
(4.3)

If *X* has a representation as in (4.3) and  $\mathfrak{L}_{\Omega}(Y) \succeq 0$ , then

$$\mathfrak{L}_X(Y) = \sum_j (W_j \otimes I)^* \mathfrak{L}_{\Omega'}(Y) (W_j \otimes I).$$

On the other hand,  $\mathfrak{L}_{\Omega'}(Y) = \mathfrak{L}_{\Omega}(Y) \oplus I \succeq 0$ . Hence  $X \in \mathcal{D}_{\mathfrak{L}_{\Omega}}^{\circ}$ .

*Proof of Theorem* 4.6. Suppose first (4.2) holds for some  $\Omega \in \mathbb{S}_n^g$ . Since  $\mathcal{D}_{\mathfrak{L}_\Omega}^\circ = \mathcal{K}$  and evidently  $\Omega \in \mathcal{D}_{\mathfrak{L}_\Omega}^\circ$ , it follows that  $\Omega \in \mathcal{K}$ . Since 0 is assumed to be in the interior of  $\mathcal{K}$ ,

its polar dual  $\mathcal{K}^{\circ} = \mathcal{D}_{\mathfrak{L}_{\Omega}}$  is bounded by Proposition 4.3. Thus, if  $X \in \mathcal{K} = \mathcal{D}_{\mathfrak{L}_{\Omega}}^{\circ}$ , then by Lemma 4.7, *X* has a representation as in (4.1).

Conversely, assume that  $\Omega \in \mathcal{K}$  has the property that any  $X \in \mathcal{K}$  can be represented as in (4.1). Consider the matrix convex set

$$\Gamma = \{ V^*(I_\mu \otimes \Omega) V : \mu \in \mathbb{N}, \ V^* V = I \}.$$

Since  $\Omega \in \mathcal{K}$ , it follows that  $\Gamma \subseteq \mathcal{K}$ . On the other hand, the hypothesis is that  $\mathcal{K} \subseteq \Gamma$ . Hence  $\mathcal{K} = \Gamma$ . Now, for  $\mathfrak{L}_X$  a monic linear pencil,  $\mathfrak{L}_X(\Omega) \succeq 0$  if and only if

$$\mathfrak{L}_X(V^*(I_\mu \otimes \Omega)V) = (V \otimes I)^* \mathfrak{L}_X(I_\mu \otimes \Omega) (V \otimes I) \succeq 0$$

for all choices of  $\mu$  and isometries V. Thus,  $X \in \mathcal{K}^{\circ}$  if and only if  $\mathfrak{L}_X(\Omega) \succeq 0$ . On the other hand,  $\mathfrak{L}_X(\Omega)$  is unitarily equivalent to  $\mathfrak{L}_{\Omega}(X)$ . Thus  $X \in \mathcal{K}^{\circ}$  if and only if  $X \in \mathcal{D}_{\mathfrak{L}_{\Omega}}$ .

**Proposition 4.9.** Suppose  $\mathcal{K}$  is a closed matrix convex set containing 0. If there is an  $\Omega \in \mathcal{K}$  such that for each  $X \in \mathcal{K}$  there is a  $\mu \in \mathbb{N}$  and an isometry V such that

$$X_j = V^* (I_\mu \otimes \Omega'_j) V, \tag{4.4}$$

then

$$\mathcal{K}^{\circ} = \mathcal{D}_{\mathfrak{L}_{\Omega}},\tag{4.5}$$

where  $\mathfrak{L}_{\Omega}$  is the monic linear pencil  $\mathfrak{L}_{\Omega}(x) = I - \sum \Omega_j x_j$ . Here  $\Omega' = \Omega \oplus 0$  as in Lemma 4.7.

Conversely, if there is an  $\Omega$  such that (4.5) holds, then  $\Omega \in \mathcal{K}$  and for each  $X \in \mathcal{K}$  there is an isometry V such that (4.4) holds.

*Proof.* Suppose first (4.5) holds for some  $\Omega \in \mathbb{S}_n^g$ . Since  $\mathcal{D}_{\mathfrak{L}_\Omega}^\circ = \mathcal{K}$  and evidently  $\Omega \in \mathcal{D}_{\mathfrak{L}_\Omega}^\circ$ , it follows that  $\Omega \in \mathcal{K}$ . By Lemma 4.7, if  $X \in \mathcal{K} = \mathcal{D}_{\mathfrak{L}_\Omega}^\circ$ , then X has a representation as in (4.4).

Conversely, assume that  $\Omega$  has the property that any  $X \in \mathcal{K}$  can be represented as in (4.4). Consider the matrix convex set

$$\Gamma = \{ V^*(I_\mu \otimes \Omega') V : \mu \in \mathbb{N}, \ V^* V = I \}.$$

Since  $0, \Omega \in \mathcal{K}$ , it follows that  $\Omega' = \Omega \oplus 0 \in \mathcal{K}$ , and thus  $\Gamma \subseteq \mathcal{K}$ . On the other hand, the hypothesis is that  $\mathcal{K} \subseteq \Gamma$ . Hence  $\mathcal{K} = \Gamma$ . Now, for  $\mathfrak{L}_X$  a monic linear pencil,  $\mathfrak{L}_X(\Omega) \succeq 0$  if and only if  $\mathfrak{L}_X(\Omega') \succeq 0$ , if and only if

$$\mathfrak{L}_X(V^*(I_\mu \otimes \Omega')V) = (V \otimes I)^* \mathfrak{L}_X(I_\mu \otimes \Omega')(V \otimes I) \succeq 0$$

for all choices of  $\mu$  and isometries V. Thus,  $X \in \mathcal{K}^{\circ}$  if and only if  $\mathfrak{L}_X(\Omega) \succeq 0$ . On the other hand,  $\mathfrak{L}_X(\Omega)$  is unitarily equivalent to  $\mathfrak{L}_{\Omega}(X)$ . Thus  $X \in \mathcal{K}^{\circ}$  if and only if  $X \in \mathcal{D}_{\mathfrak{L}_{\Omega}}$ . **Remark 4.10.** (1) For perspective, in the classical (not free) situation, when g = 2 it is known that  $K \subseteq \mathbb{R}^2$  has an LMI representation if and only if  $K^\circ$  is a numerical range [Hen10, HS12]. It is well known that the polar dual of a spectrahedron is not necessarily a spectrahedron. This is the case even in  $\mathbb{R}^g$  (cf. [BPR13, Section 5] or Example 8.4).

(2) In the commutative case the polar dual of a spectrahedron (more generally, of a spectrahedral shadow) is a spectrahedral shadow (see [GN11] or [BPR13, Chapter 5]).

(3) It turns out that the  $\Omega$  in Theorem 4.6 can be taken to be an extreme point of  $\mathcal{K}$  in a very strong free sense. We refer to [Far00, Kls14, WW99] for more on matrix extreme points.

#### 4.4. The polar dual of a free spectrahedrop is a free spectrahedrop

This subsection contains a duality result for free spectrahedrops (Theorem 4.11) and several of its corollaries.

It can happen that  $\mathcal{D}_{\mathfrak{L}}$  is not bounded, but the projection  $\mathcal{K} = \operatorname{proj}_{\mathfrak{X}} \mathcal{D}_{\mathfrak{L}}$  is. Corollary 4.13 below says that a free spectrahedrop is closed and bounded if and only if it is the projection of some bounded free spectrahedron. For expositional purposes, it is convenient to introduce the following terminology. A free spectrahedrop  $\mathcal{K}$  is called *stratospherically bounded* if there is a linear pencil  $\mathfrak{L}$  such that  $\mathcal{K} = \operatorname{proj}_{\mathfrak{X}} \mathcal{D}_{\mathfrak{L}}$  and  $\mathcal{D}_{\mathfrak{L}}$  is bounded.

**Theorem 4.11.** Suppose  $\mathcal{K}$  is a closed matrix convex set containing 0.

- (1) If  $\mathcal{K}$  is a free spectrahedrop and 0 is in the interior of  $\mathcal{K}$ , then  $\mathcal{K}^{\circ}$  is a stratospherically bounded free spectrahedrop.
- (2) If  $\mathcal{K}^{\circ}$  is a free spectrahedrop containing 0 in its interior, then  $\mathcal{K}$  is a stratospherically bounded free spectrahedrop.

In particular, if  $\mathcal{K}$  is a bounded free spectrahedrop with 0 in its interior, then both  $\mathcal{K}$  and  $\mathcal{K}^{\circ}$  are stratospherically bounded free spectrahedrops (with 0 in their interiors).

Before presenting the proof of the theorem we state a few corollaries and Proposition 4.14 needed in the proof.

**Corollary 4.12.** Given  $\Omega \in \mathbb{S}_d^g$ , let  $\mathfrak{L}_\Omega$  denote the corresponding monic linear pencil. The free set  $\mathcal{D}_{\mathfrak{L}_\Omega}^\circ$  is a stratospherically bounded free spectrahedrop.

*Proof.* The set  $\mathcal{D}_{\mathfrak{L}_{\Omega}}$  is (trivially) a free spectrahedrop with 0 in its interior. Thus, by Theorem 4.11,  $\mathcal{D}_{\mathfrak{L}_{\Omega}}^{\circ}$  is a stratospherically bounded free spectrahedrop.

**Corollary 4.13.** A free spectrahedrop  $\mathcal{K} \subseteq \mathbb{S}^g$  is closed and bounded if and only if it is stratospherically bounded.

*Proof.* Implication ( $\Leftarrow$ ) is obvious. ( $\Rightarrow$ ) Let us first reduce to the case where  $\mathcal{K}(1)$  has nonempty interior. If  $\mathcal{K}(1)$  has empty interior, then it is contained in a proper affine hyperplane { $\ell = 0$ } of  $\mathbb{R}^g$ . Here  $\ell$  is an affine linear functional. In this case we can solve for one of the variables, thereby reducing the codimension of  $\mathcal{K}(1)$ . (Note that  $\ell = 0$  on  $\mathcal{K}(1)$  implies  $\ell = 0$  on  $\mathcal{K}$  [HKM16, Lemma 3.3].)

Now let  $\hat{x} \in \mathbb{R}^g$  be an interior point of  $\mathcal{K}(1)$ . Consider the translation

$$\tilde{\mathcal{K}} = \mathcal{K} - \hat{x} = \bigcup_{n \in \mathbb{N}} \{ X - \hat{x} I_n : X \in \mathcal{K}(n) \}.$$
(4.6)

Clearly,  $\tilde{\mathcal{K}}$  is a bounded free spectrahedrop with 0 in its interior. Hence by Theorem 4.11, it is stratospherically bounded. Translating back, we see  $\mathcal{K}$  is a stratospherically bounded free spectrahedrop. 

Each stratospherically bounded free spectrahedrop is closed, since it is the projection of a (levelwise) compact spectrahedron. Hence a bounded free spectrahedrop  $\mathcal{K}$  will not be stratospherically bounded if it is not closed. For a concrete example, consider the linear pencil

$$L(x, y) = \begin{pmatrix} 2-x & 1\\ 1 & 2-y \end{pmatrix} \oplus (2+x),$$

and let  $\mathcal{K} = \operatorname{proj}_{X} \mathcal{D}_{L}$ . Then  $\mathcal{K} = \{X \in \mathbb{S} : -2 \leq X < 2\}$  is bounded but not closed.

**Proposition 4.14.** Given  $\Omega \in \mathbb{S}_d^g$  and  $\Gamma \in \mathbb{S}_d^h$ , the sequence  $\mathcal{K} = (\mathcal{K}(n))_n$ ,

$$\mathcal{K}(n) = \{ A \in \mathbb{S}_n^g : A = V^*(I_\mu \otimes \Omega)V, \\ 0 = V^*(I_\mu \otimes \Gamma)V \text{ for some isometry } V \text{ and } \mu \le nd \}$$

is a stratospherically bounded free spectrahedrop.

Let  $\mathfrak{L}_{(\Omega,\Gamma)}$  denote the monic linear pencil corresponding to  $(\Omega,\Gamma)$ . The free set

$$\mathcal{C} = \{A : (A, 0) \in \mathcal{D}^{\circ}_{\mathfrak{L}(\Omega, \Gamma)}\}$$

is a stratospherically bounded free spectrahedrop.

*Proof.* Let S denote the span of  $\{I, \Omega_1, \ldots, \Omega_g, \Gamma_1, \ldots, \Gamma_h\}$ . Thus S is an operator system in  $M_d$  (the fact that  $I \in S$  implies S contains a positive definite element). Let  $\phi: \mathcal{S} \to M_n$  denote the linear mapping determined by

$$I \mapsto I, \quad \Omega_j \mapsto A_j, \quad \Gamma_\ell \mapsto 0.$$

Observe that, by Theorem 3.1,  $A \in \mathcal{K}(n)$  if and only if  $\phi$  has a completely positive extension  $\Phi: M_d \to M_n$ . Theorem 3.4 expresses existence of such a  $\Phi$  as a (unital) cp interpolation problem in terms of a free spectrahedron. For the reader's convenience we write out this critical LMI explicitly. Let  $\omega_{pq}^{j}$  denote the (p, q)-entry of  $\Omega_{j}$  and  $\gamma_{pq}^{\ell}$  the (p, q)-entry of  $\Gamma_{\ell}$ . For a complex matrix (or scalar) Q we use  $\hat{Q}$  to denote its real part and  $i\dot{Q}$  for its imaginary part. Thus  $\hat{Q} = \frac{1}{2}(Q+Q^*)$  and  $\dot{Q} = -\frac{i}{2}(Q-Q^*)$ .

Now A is in  $\mathcal{K}(n)$  if and only if there exist  $n \times n$  matrices  $C_{p,q}$  satisfying

- (i)  $\sum_{p,q=1}^{d} E_{p,q} \otimes C_{p,q} \succeq 0;$ (ii)  $\sum_{p=1}^{d} C_{p,p} = I_n;$ (iii)  $\sum_{p,q=1}^{d} \omega_{pq}^{\ell} C_{p,q} = A_{\ell} \text{ for } \ell = 1, \dots, g; \text{ and}$ (iv)  $\sum_{p,q=1}^{d} \gamma_{pq}^{\ell} C_{p,q} = 0 \text{ for } \ell = 1, \dots, h.$

Since the  $C_{p,q}$  for  $p \neq q$  are not hermitian matrices, we rewrite the system (i)–(iv) into one with hermitian unknowns  $\hat{C}_{p,q}$  and  $\check{C}_{p,q}$ . Property (i) transforms into

$$\sum_{p,q} (\hat{E}_{p,q} \otimes \hat{C}_{p,q} - \check{E}_{p,q} \otimes \check{C}_{p,q}) + i(\hat{E}_{p,q} \otimes \check{C}_{p,q} + \check{E}_{p,q} \otimes \hat{C}_{p,q}) \succeq 0,$$

i.e.,

$$\sum_{p,q} \hat{E}_{p,q} \otimes \hat{C}_{p,q} - \check{E}_{p,q} \otimes \check{C}_{p,q} \ge 0,$$

$$\sum_{p,q} \hat{E}_{p,q} \otimes \check{C}_{p,q} + \check{E}_{p,q} \otimes \hat{C}_{p,q} = 0.$$
(4.7)

In (ii) we simply replace  $C_{p,p}$  with  $\hat{C}_{p,p}$ ,

$$\sum_{p=1}^{d} \hat{C}_{p,p} = I_n.$$
(4.8)

Properties (iii) and (iv) are handled similarly to (i). Thus

$$\sum_{p,q} \hat{\omega}_{pq}^{\ell} \hat{C}_{p,q} - \check{\omega}_{pq}^{\ell} \check{C}_{p,q} = A_{\ell},$$

$$\sum_{p,q} \hat{\omega}_{pq}^{\ell} \check{C}_{p,q} + \check{\omega}_{pq}^{\ell} \hat{C}_{p,q} = 0,$$
(4.9)

and

$$\sum_{p,q} \hat{\gamma}_{pq}^{\ell} \hat{C}_{p,q} - \check{\gamma}_{pq}^{\ell} \check{C}_{p,q} = A_{\ell},$$

$$\sum_{p,q} \hat{\gamma}_{pq}^{\ell} \check{C}_{p,q} + \check{\gamma}_{pq}^{\ell} \hat{C}_{p,q} = 0.$$
(4.10)

Hence,  $\mathcal{K}$  is the linear image of the explicitly constructed free spectrahedron (in the variables  $\hat{C}_{p,q}$  and  $\check{C}_{p,q}$ ) given by (4.7)–(4.10). Moreover, (i) and (ii) together imply  $0 \leq C_{p,p} \leq I$ . It now follows that  $\|\hat{C}_{p,q}\|, \|\check{C}_{p,q}\| \leq 1$  for all p, q. Thus, this free spectrahedron is bounded. It is now routine to verify that  $\mathcal{K}$  is a projection of a bounded free spectrahedron and is thus a stratospherically bounded free spectrahedrop.

Let  $\Omega' = \Omega \oplus 0$  and  $\Gamma' = \Gamma \oplus 0$  where  $0 \in \mathbb{S}_d^g$ . Note that

$$\mathcal{D}_{\mathfrak{L}_{(\Omega,\Gamma)}} = \mathcal{D}_{\mathfrak{L}_{(\Omega',\Gamma')}}.$$

By Lemma 4.7,  $(A, 0) \in \mathcal{D}^{\circ}_{\mathfrak{L}(\Omega, \Gamma)}$  if and only if

 $A \in \{B : \exists \mu \in \mathbb{N} \text{ and an isometry } V \text{ such that } B = V^*(I_\mu \otimes \Omega')V, \ 0 = V^*(I_\mu \otimes \Gamma')V\}.$ 

By the first part of the proposition (applied to the tuple  $(\Omega', \Gamma')$ ), it follows that C is a stratospherically bounded free spectrahedrop.

*Proof of Theorem 4.11.* Suppose  $\mathcal{K}$  is a free spectrahedrop with 0 in its interior. By Lemma 4.1, there exists  $(\Omega, \Gamma)$ , a pair of tuples of matrices, such that

$$\mathcal{K} = \{X : \exists Y, \ (X, Y) \in \mathcal{D}_{\mathfrak{L}_{(\Omega, \Gamma)}}\} = \operatorname{proj}_{x} \mathcal{D}_{\mathfrak{L}_{(\Omega, \Gamma)}},$$

where  $\mathfrak{L}_{(\Omega,\Gamma)}(x, y)$  is the monic linear pencil associated to  $(\Omega, \Gamma)$ .

Observe that  $A \in \mathcal{K}^{\circ}$  if and only if

$$\mathfrak{L}_A(X) \succeq 0$$
 for each  $X \in \mathcal{K}$ .

Thus,  $A \in \mathcal{K}^{\circ}$  if and only if

$$\mathfrak{L}_{(A,0)}(X,Y) \succeq 0 \quad \text{for all } (X,Y) \in \mathcal{D}_{\mathfrak{L}_{(\Omega,\Gamma)}},$$

if and only if

$$(A, 0) \in \mathcal{D}^{\circ}_{\mathfrak{L}(\Omega, \Gamma)}$$

Summarizing,  $A \in \mathcal{K}^{\circ}$  if and only if  $(A, 0) \in \mathcal{D}^{\circ}_{\mathfrak{L}_{(\Omega,\Gamma)}}$ . Thus, by the second part of Proposition 4.14,  $\mathcal{K}^{\circ}$  is a stratospherically bounded free spectrahedrop.

Because  $\mathcal{K}$  contains 0 and is a closed matrix convex set,  $\mathcal{K}^{\circ\circ} = \mathcal{K}$  by Proposition 4.3. Thus, if  $\mathcal{K}^{\circ}$  is a free spectrahedrop with 0 in its interior, then, by what has already been proved,  $\mathcal{K}^{\circ\circ} = \mathcal{K}$  is a stratospherically bounded free spectrahedrop.

Finally, if  $\mathcal{K}$  is a bounded free spectrahedrop with 0 in its interior, then  $\mathcal{K}^{\circ}$  contains 0 in its interior and is a stratospherically bounded free spectrahedrop. Hence,  $\mathcal{K} = \mathcal{K}^{\circ\circ}$  is also a stratospherically bounded free spectrahedrop.

Note that the polar dual of a free spectrahedron is a matrix convex set generated by a singleton (Theorem 4.6) and is a free spectrahedrop by the above corollary.

**Corollary 4.15.** Let  $\mathfrak{L}$  denote the monic linear pencil associated with  $(\Omega, \Gamma)$ . If  $\mathcal{K} = \text{proj}_x \mathcal{D}_{\mathfrak{L}}$  is bounded, then its polar dual is the free set given by

$$\mathcal{K}^{\circ}(n) = \{ A \in \mathbb{S}_{n}^{g} : (A, 0) \in \mathcal{D}_{\mathfrak{L}}^{\circ} \}$$
$$= \{ A \in \mathbb{S}_{n}^{g} : \exists \mu \in \mathbb{N} \text{ and an isometry } V \text{ with } A = V^{*}(I_{\mu} \otimes \Omega)V, \ 0 = V^{*}(I_{\mu} \otimes \Gamma)V \}.$$

Whether or not K is bounded, its polar dual is the free set

 $\mathcal{K}^{\circ}(n) = \{A \in \mathbb{S}_{n}^{g} : (A, 0) \in \mathcal{D}_{\mathfrak{L}}^{\circ}\}\$ =  $\{A \in \mathbb{S}_{n}^{g} : \exists \mu \in \mathbb{N} \text{ and an isometry } V \text{ with } A = V^{*}(I_{\mu} \otimes \Omega')V, 0 = V^{*}(I_{\mu} \otimes \Gamma')V\},\$ 

where  $\Omega' = \Omega \oplus 0$  and  $\Gamma' = \Gamma \oplus 0$ , as in Lemma 4.7.

*Proof.* From the proof of Theorem 4.11,  $\mathcal{K}^{\circ} = \{A : (A, 0) \in \mathcal{D}_{\mathfrak{L}}^{\circ}\}$ . Writing  $\mathfrak{L} = \mathfrak{L}_{\Delta}$ , by Lemma 4.7 (whether or not  $\mathcal{D}_{\mathfrak{L}}$  is bounded),

 $\mathcal{D}_{\mathfrak{L}}^{\circ} = \{X : \exists \mu \in \mathbb{N} \text{ and an isometry } V \text{ such that } X = V^*(I_{\mu} \otimes \Delta')V\}.$ 

To obtain the stronger conclusion under the assumption that  $\mathcal{K}$  is bounded, an additional argument along the lines of [HKM13a, §3.1] is needed; see also [Zal17, Theorem 2.12]. Let  $(A, 0) \in \mathcal{D}^{\circ}_{\mathcal{L}}$ . We need to show that the unital linear map

$$\tau : \operatorname{span}\{I, \Omega_1, \dots, \Omega_g, \Gamma_1, \dots, \Gamma_h\} \to \operatorname{span}\{I, A_1, \dots, A_g\},$$
$$\Gamma_j \mapsto A_j, \quad \Gamma_k \mapsto 0,$$

is completely positive. Assume

$$I \otimes X_0 + \sum_j \Omega_j \otimes X_j + \sum_k \Gamma_k \otimes Y_k \succeq 0$$
(4.11)

for some hermitian  $X_0, \ldots, X_g, Y_1, \ldots, Y_h$ . In particular,  $X_0 = X_0^*$ . We claim that  $X_0 \succeq 0$ . Suppose  $X_0 \not\succeq 0$ . By compressing we may reduce to  $X_0 \prec 0$ . From (4.11) it now follows that

$$I \otimes tX_0 + \sum_j \Omega_j \otimes tX_j + \sum_k \Gamma_k \otimes tY_k \succeq 0$$

for every t > 0. Since  $tX_0 \prec 0$ , this implies

$$I \otimes I + \sum_{j} \Omega_{j} \otimes tX_{j} + \sum_{k} \Gamma_{k} \otimes tY_{k} \succeq 0,$$

whence  $(tX_1, \ldots, tX_g) \in \mathcal{K}$  for every t > 0. If  $(X_1, \ldots, X_g) \neq 0$ , this contradicts the boundedness of  $\mathcal{K}$ . Otherwise  $(X_1, \ldots, X_g) = 0$ , and

$$\sum_{k} \Gamma_k \otimes Y_k \succ -I \otimes X_0 \succ 0.$$

Hence for any tuple  $(X_1, \ldots, X_g)$  of hermitian matrices of the same size as the  $Y_k$ ,

$$I \otimes I + \sum_{j} \Omega_{j} \otimes X_{j} + \sum_{k} \Gamma_{k} \otimes tY_{k} \succeq 0$$

for some t > 0. This again contradicts the boundedness of  $\mathcal{K}$ . Thus  $X_0 \succeq 0$ .

By adding a small multiple of the identity to  $X_0$  there is no harm in assuming  $X_0 \succ 0$ . Hence multiplying (4.11) by  $X_0^{-1/2}$  on the left and right yields the tuple  $X_0^{-1/2}(X_1, \ldots, X_g, Y_1, \ldots, Y_h)X_0^{-1/2} \in \mathcal{D}_{\mathcal{L}}$ . Since  $(A, 0) \in \mathcal{D}_{\mathcal{L}}^{\circ}$ ,

$$\mathcal{L}_{(A,0)}(X_0^{-1/2}(X_1,\ldots,X_g,Y_1,\ldots,Y_h)X_0^{-1/2}) = I \otimes I + \sum_k A_k \otimes X_0^{-1/2}X_k X_0^{-1/2} \succeq 0.$$

Multiplying by  $X_0^{1/2}$  on the left and right gives

$$I\otimes X_0+\sum_k A_k\otimes X_k\succeq 0,$$

as required.

To each subset  $\Gamma \subseteq \mathbb{S}^g$  we associate its interior int  $\Gamma = (\operatorname{int} \Gamma(n))_{n \in \mathbb{N}}$ , where int  $\Gamma(n)$  denotes the interior of  $\Gamma(n)$  in the Euclidean space  $\mathbb{S}_n^g$ . We say  $\Gamma$  has nonempty interior if there is *n* with int  $\Gamma(n) \neq \emptyset$ .

**Corollary 4.16.** If  $\mathcal{K} \subseteq \mathbb{S}^g$  is a bounded free spectrahedrop, then  $\overline{\mathcal{K}}$  is a free spectrahedrop.

*Proof.* As in the proof of Corollary 4.13, we may assume the interior of  $\mathcal{K}$  is nonempty. This implies there is an  $\hat{x} \in \mathbb{R}^g$  in the interior of  $\mathcal{K}(1)$ . Consider the translation  $\tilde{\mathcal{K}} = \mathcal{K} - \hat{x}$  as in (4.6). This is a free spectrahedrop containing 0 in its interior. Hence its closure  $\tilde{\mathcal{K}} = \tilde{\mathcal{K}}^{\circ\circ}$  is a free spectrahedrop by Theorem 4.11. Thus so is  $\overline{\mathcal{K}} = \tilde{\mathcal{K}} + \hat{x}$ .

**Corollary 4.17.** If  $\mathcal{K} \subseteq \mathbb{S}^g$  is a free spectrahedrop with nonempty interior, then  $\mathcal{K}^\circ$  is a free spectrahedrop.

*Proof.* Assume  $\mathcal{K} = \text{proj } \mathcal{D}_{L_A}$ . Applying Lemma 4.5 gives

$$\mathcal{K}^{\circ} = \{ B \in \mathbb{S}^g : (B, 0) \in \mathcal{D}_{L_A}^{\circ} \}.$$

So if we prove  $\mathcal{D}_{L_4}^{\circ}$  is a free spectrahedrop, then

$$\mathcal{K}^{\circ} = \operatorname{proj} \left( \mathcal{D}_{L_{A}}^{\circ} \cap (\mathbb{S}^{g} \otimes \{0\}^{h}) \right)$$

is the intersection of two free spectrahedrops, so a free spectrahedrop. Thus without loss of generality we may take  $\mathcal{K} = \mathcal{D}_{L_A}$  and proceed. We will demonstrate that the corollary in this case is a consequence of the Convex Positivstellensatz, Theorem 2.4.

Suppose  $\hat{x} \in \mathbb{R}^g$  is in the interior of  $\mathcal{D}_{L_A}$ . Without loss of generality we may assume

$$L_0 = L_A(\hat{x}) \succ 0$$

(cf. [HV07]). Define the monic linear pencil

$$\mathfrak{L}(\mathbf{y}) = L_0^{-1/2} L(\mathbf{y} + \hat{x}) L_0^{-1/2} = I + \sum_{j=1}^g L_0^{-1/2} A_j L_0^{-1/2} y_j$$

By definition, a tuple  $\Omega \in \mathbb{S}^g$  is in  $\mathcal{D}_{L_A}^{\circ}$  if and only if  $\mathcal{D}_{L_A} \subseteq \mathcal{D}_{\mathfrak{L}_\Omega}$ . Equivalently, with

$$L(y) = \left(I + \sum_{j=1}^{g} \Omega_j \hat{x}_j\right) + \sum_{j=1}^{g} \Omega_j y_j,$$

we have  $\mathcal{D}_{\mathfrak{L}} \subseteq \mathcal{D}_L$ . By Theorem 2.4, there is an  $S \succeq 0$  and matrices  $V_k$  with

$$L(y) = S + \sum_{k} V_{k}^{*} \mathfrak{L}(y) V_{k}.$$

That is,

$$I + \sum_{j=1}^{g} \Omega_j \hat{x}_j \succeq \sum_k V_k^* V_k, \text{ and } \Omega_j = \sum_k V_k^* L_0^{-1/2} A_j L_0^{-1/2} V_k, \quad j = 1, \dots, g.$$

Equivalently, there is a completely positive mapping  $\Phi$  satisfying

$$\Phi(L_0^{-1/2}A_jL_0^{-1/2}) = \Omega_j, \quad j = 1, \dots, g, \tag{4.12}$$

$$\Phi(I) \leq I + \sum_{j} \Omega_j \hat{x}_j. \tag{4.13}$$

As in Theorem 3.4 we now employ the Choi matrix C. Conditions (4.12) translate into linear constraints on the block entries  $C_{ij}$  of C. Similarly, (4.13) transforms into an LMI constraint on the entries of C. Thus C provides a free spectrahedral lift of  $\mathcal{D}_{L_A}^\circ$ .

## 4.5. The free convex hull of a union

In this subsection we prove that the convex hull of a union of free spectrahedrops is again a free spectrahedrop.

**Proposition 4.18.** If  $S_1, \ldots, S_t \subseteq \mathbb{S}^g$  are stratospherically bounded free spectrahedrops and each contains 0 in its interior, then  $\overline{co}^{mat}(S_1 \cup \cdots \cup S_t)$  is a stratospherically bounded free spectrahedrop with 0 in its interior.

*Proof.* Let  $\mathcal{K} = \mathcal{S}_1 \cup \cdots \cup \mathcal{S}_t$ . Then

$$\mathcal{K}^{\circ} = \mathcal{S}_{1}^{\circ} \cap \cdots \cap \mathcal{S}_{t}^{\circ}$$

Since each  $S_j$  is a stratospherically bounded free spectrahedrop with 0 in its interior, the same holds true for  $S_j^{\circ}$  by Theorem 4.11. It is clear that these properties are preserved under a finite intersection, so  $\mathcal{K}^{\circ}$  is again a stratospherically bounded free spectrahedrop with 0 in its interior. Hence, by Proposition 4.3,

$$(\mathcal{K}^{\circ})^{\circ} = \overline{\mathrm{co}}^{\mathrm{mat}} \mathcal{K}$$

is a stratospherically bounded free spectrahedrop with 0 in its interior by Theorem 4.11.

## 5. Positivstellensatz for free spectrahedrops

This section focuses on polynomials positive on a free spectrahedrop, extending our Positivstellensatz for free polynomials positive on free spectrahedra, Theorem 2.4, to a Convex Positivstellensatz for free spectrahedrops, Theorem 5.1.

Let  $\mathfrak{L}$  denote a monic linear pencil of size d,

$$\mathfrak{L}(x, y) = I + \sum_{j=1}^{g} \Omega_j x_j + \sum_{k=1}^{h} \Gamma_k y_k,$$
(5.1)

and let  $\mathcal{K} = \operatorname{proj}_x \mathcal{D}_{\mathfrak{L}}$ . If  $\mu$  is a positive integer and  $q_{\ell} \in \mathbb{C}^{d \times \mu} \langle x \rangle$  (and so are polynomials in the *x* variables only), and if  $\sum_{\ell} q_{\ell}(x)^* \Gamma_k q_{\ell}(x) = 0$  for each *k*, then

$$\sum_{\ell} q_{\ell}^*(x) \mathfrak{L}(x, y) q_{\ell}(x)$$

is a polynomial in the x variables and is thus in  $\mathbb{C}^{\mu \times \mu} \langle x \rangle$ . For positive integers  $\mu$  and r we define the *truncated quadratic module* in  $\mathbb{C}^{\mu \times \mu} \langle x \rangle$  associated to  $\mathfrak{L}$  and  $\mathcal{K}$  by

$$M_x^{\mu}(\mathfrak{L})_r = \left\{ \sum_{\ell} q_{\ell}^* \mathfrak{L} q_{\ell} + \sigma : q_{\ell} \in \mathbb{C}^{d \times \mu} \langle x \rangle_r, \ \sigma \in \Sigma_r^{\mu} \langle x \rangle, \ \sum_{\ell} q_{\ell}^* \Gamma_k q_{\ell} = 0 \text{ for all } k \right\}$$

Here  $\Sigma_r^{\mu} = \Sigma_r^{\mu} \langle x \rangle$  denotes the set of all sums of hermitian squares  $h^*h$  for  $h \in \mathbb{C}^{\mu \times \mu} \langle x \rangle_r$ . It is easy to see that  $M_x^{\mu}(\mathfrak{L}) = \bigcup_{r \in \mathbb{N}} M_x^{\mu}(\mathfrak{L})_r$  is a quadratic module in  $\mathbb{C}^{\mu \times \mu} \langle x \rangle$ .

The main result of this section is the following Positivstellensatz:

**Theorem 5.1.** A symmetric polynomial  $p \in \mathbb{C}^{\mu \times \mu} \langle x \rangle_{2r+1}$  is positive semidefinite on  $\mathcal{K}$  if and only if  $p \in M_x^{\mu}(\mathfrak{L})_r$ .

**Remark 5.2.** (1) In case there are no y variables in  $\mathfrak{L}$ , Theorem 5.1 reduces to the Convex Positivstellensatz of [HKM12].

(2) If r = 0, i.e., p is linear, then Theorem 5.1 reduces to Corollary 4.15.

(3) A Positivstellensatz for commutative polynomials *strictly positive* on spectrahedrops was established by Gouveia and Netzer [GN11]. A major distinction is that the degrees of the  $q_i$  and  $\sigma$  in the commutative theorem behave very badly.

(4) Observe that  $\mathcal{K}$  is in general not closed. Thus Theorem 5.1 yields a "perfect" Positivstellensatz for certain nonclosed sets.

## 5.1. Proof of Theorem 5.1

We begin with some auxiliary results.

**Proposition 5.3.** With  $\mathfrak{L}$  a monic linear pencil as in (5.1),  $M_x^{\mu}(\mathfrak{L})_r$  is a closed convex cone in the set of all symmetric polynomials in  $\mathbb{C}^{\mu \times \mu} \langle x \rangle_{2r+1}$ .

The convex cone property is obvious. For the proof that this cone is closed, it is convenient to introduce a norm compatible with  $\mathfrak{L}$ .

Given  $\varepsilon > 0$ , let

$$\mathcal{B}^{g}_{\varepsilon}(n) := \{ X \in \mathbb{S}^{g}_{n} : \|X\| \le \varepsilon \}.$$

There is an  $\varepsilon > 0$  such that for all  $n \in \mathbb{N}$ , if  $(X, Y) \in \mathbb{S}_n^{g+h}$  and  $||(X, Y)|| \le \varepsilon$ , then  $\mathfrak{L}(X, Y) \ge 1/2$ . In particular,  $\mathcal{B}_{\varepsilon}^{g+h} \subseteq \mathcal{D}_{\mathfrak{L}}$ . Using this  $\varepsilon$  we norm matrix polynomials in g + h variables by

$$||p(x, y)|| := \max\{||p(X, Y)|| : (X, Y) \in \mathcal{B}_{\varepsilon}^{g+h}\}.$$
(5.2)

(Note that by the nonexistence of polynomial identities for matrices of all sizes, ||| p(x, y) ||| = 0 iff p(x, y) = 0 [Row80, §2.5, §1.4]. Furthermore, on the right-hand side of (5.2) the maximum is attained because the bounded free semialgebraic set  $\mathcal{B}_{\varepsilon}^{g+h}$  is levelwise compact and matrix convex; see [HM04, Section 2.3] for details.) Note that if  $f \in \mathbb{C}^{d \times \mu} \langle x \rangle_{\beta}$  and  $||| f(x)^* \mathfrak{L}(x, y) f(x) ||| \le N^2$ , then  $||| f^* f ||| \le 2N^2$ .

*Proof of Proposition 5.3.* Suppose  $(p_n)$  is a sequence from  $M_x^{\mu}(\mathfrak{L})_r$  that converges to some symmetric  $p \in \mathbb{C}^{\mu \times \mu} \langle x \rangle$  of degree at most 2r + 1. By Carathéodory's convex hull

theorem (see e.g. [Bar02, Theorem I.2.3]), there is an *M* such that for each *n* there exist matrix-valued polynomials  $r_{n,i} \in \mathbb{C}^{\mu \times \mu} \langle x \rangle_r$  and  $t_{n,i} \in \mathbb{C}^{d \times \mu} \langle x \rangle_r$  such that

$$p_n = \sum_{i=1}^{M} r_{n,i}^* r_{n,i} + \sum_{i=1}^{M} t_{n,i}^* \mathfrak{L}(x, y) t_{n,i}$$

Since  $|||p_n||| \le N^2$ , it follows that  $|||r_{n,i}||| \le N$  and likewise  $|||t_{n,i}^* \mathcal{L}(x, y)t_{n,i}||| \le N^2$ . In view of the remarks preceding the proof, we obtain  $|||t_{n,i}||| \le \sqrt{2}N$  for all *i*, *n*. Hence for each *i*, the sequences  $(r_{n,i})$  and  $(t_{n,i})$  are bounded in *n*. They thus have convergent subsequences. Passing to one of these subsequential limits finishes the proof.

Next is a variant of the Gelfand-Naimark-Segal (GNS) construction.

**Proposition 5.4.** If  $\lambda : \mathbb{C}^{\nu \times \nu} \langle x \rangle_{2k+2} \to \mathbb{C}$  is a linear functional that is nonnegative on  $\Sigma_{k+1}^{\nu}$  and positive on  $\Sigma_{k}^{\nu} \setminus \{0\}$ , then there exists a tuple  $X = (X_1, \ldots, X_g)$  of hermitian operators on a Hilbert space  $\mathcal{X}$  of dimension at most  $\nu \dim \mathbb{C} \langle x \rangle_k$  and a vector  $\gamma \in \mathcal{X}^{\oplus \nu}$  such that

$$\lambda(f) = \langle f(X)\gamma, \gamma \rangle \tag{5.3}$$

for all  $f \in \mathbb{C}^{\nu \times \nu} \langle x \rangle_{2k+1}$ , where  $\langle \Box, \Box \rangle$  is the inner product on  $\mathcal{X}$ . Further, if  $\lambda$  is nonnegative on  $M_x^{\nu}(\mathfrak{L})_k$ , then X is in the closure  $\overline{\mathcal{K}}$  of the free spectrahedrop  $\mathcal{K}$  coming from  $\mathfrak{L}$ .

Conversely, if  $X = (X_1, ..., X_g)$  is a tuple of symmetric operators on a Hilbert space  $\mathcal{X}$  of dimension N, the vector  $\gamma \in \mathcal{X}^{\oplus \nu}$ , and k is a positive integer, then the linear functional  $\lambda : \mathbb{C}^{\nu \times \nu} \langle x \rangle_{2k+2} \to \mathbb{C}$  defined by

$$\lambda(f) = \langle f(X)\gamma, \gamma \rangle$$

is nonnegative on  $\Sigma_{k+1}^{\nu}$ . Further, if  $X \in \overline{\mathcal{K}}$ , then  $\lambda$  is nonnegative also on  $M_x^{\nu}(\mathfrak{L})_k$ .

*Proof.* The first part of the forward direction is standard: see e.g. [HKM12, Proposition 2.5]. In the course of the proof one constructs  $X_j$  as the operators of multiplication by  $x_j$  on a Hilbert space  $\mathcal{X}$ , that, as a set, is  $\mathbb{C}\langle x \rangle_k^{1 \times \nu}$  (the set of row vectors of length  $\nu$  whose entries are polynomials of degree at most k). The vector space  $\mathcal{X}^{\oplus \nu}$  in which  $\gamma$  lies is  $\mathbb{C}\langle x \rangle_k^{\nu \times \nu}$  and  $\gamma$  can be thought of as the identity matrix in  $\mathbb{C}\langle x \rangle_k^{\nu \times \nu}$ . Indeed, the (column) vector  $\gamma$  has *j*-th entry the row vector with *j*-th entry the empty set (which plays the role of multiplicative identity) and zeros elsewhere.

In particular, for  $p \in \mathcal{X} = \mathbb{C}\langle x \rangle_k^{1 \times \nu}$ , we have  $p = p(X)\gamma$ . Let  $\sigma$  denote the dimension of  $\mathcal{X}$  (which turns out to be  $\nu$  times the dimension of  $\mathbb{C}\langle x \rangle_k$ ).

We next assume that  $\lambda$  is nonnegative on  $M_x^{\nu}(\mathfrak{L})_k$  and claim that then  $X \in \overline{\mathcal{K}}$ . Assume otherwise. Then, as  $\overline{\mathcal{K}}$  is closed matrix convex (and  $\mathcal{K}$  contains 0 since  $\mathfrak{L}$  is monic), the matricial Hahn–Banach Theorem 2.2 applies: there is a monic linear pencil  $\mathfrak{L}_{\Lambda}$  of size  $\sigma$  such that  $\mathfrak{L}_{\Lambda}|_{\mathcal{K}} \succeq 0$  and  $\mathfrak{L}_{\Lambda}(X) \not\succeq 0$ . In particular,  $\mathcal{D}_{\mathfrak{L}_{\Lambda}} \supseteq \mathcal{K}$ , whence

$$\mathcal{D}^{\circ}_{\mathfrak{L}_{\Lambda}} \subseteq \mathcal{K}^{\circ}.$$

By Corollary 4.15,

$$\mathcal{K}^{\circ}(n) = \left\{ A \in \mathbb{S}_{n}^{g} : \exists \mu \in \mathbb{N} \text{ } \exists \text{ isometry } V, \ \sum_{j=1}^{\mu} V_{j}^{*} \Gamma V_{j} = 0, \ \sum_{j=1}^{\mu} V_{j}^{*} \Omega V_{j} = A \right\}.$$
(5.4)

Since  $\Lambda \in \mathcal{K}^{\circ}$ , there is an isometry *W* with

$$\sum_{j=1}^{\eta} W_j^* \Gamma W_j = 0, \qquad \sum_{j=1}^{\eta} W_j^* \Omega W_j = \Lambda.$$

Here,  $W = \operatorname{col}(W_1, \ldots, W_\eta)$  for some  $\eta$ , and  $W_j \in \mathbb{C}^{d \times \sigma}$ . Since  $\mathfrak{L}_{\Lambda}(X) \not\geq 0$ , there is  $u \in \mathbb{C}^{\sigma} \otimes \mathcal{X}$  with

$$u^* L_\Lambda(X) u < 0. \tag{5.5}$$

Let

$$u=\sum_i e_i\otimes v_i,$$

where  $e_i \in \mathbb{C}^{\sigma}$  are the standard basis vectors, and  $v_i \in \mathcal{X}$ . By the construction of X and  $\gamma$ , there is a polynomial  $p_i \in \mathbb{C}\langle x \rangle_k^{1 \times \nu}$  with  $v_i = p_i(X)\gamma$ . Now (5.5) can be written as follows:

$$0 > u^{*} \mathfrak{L}_{\Lambda}(X) u = \left(\sum_{i} e_{i} \otimes v_{i}\right)^{*} \mathfrak{L}_{\Lambda}(X) \left(\sum_{j} e_{j} \otimes v_{j}\right)$$
$$= \sum_{i, j, \ell} (e_{i} \otimes v_{i})^{*} (W_{\ell} \otimes I)^{*} \mathfrak{L}(X, Y) (W_{\ell} \otimes I) (e_{j} \otimes v_{j})$$
$$= \sum_{i, j, \ell} (W_{\ell} e_{i} \otimes p_{i}(X) \gamma)^{*} \mathfrak{L}(X, Y) (W_{\ell} e_{j} \otimes p_{j}(X) \gamma).$$
(5.6)

If we let  $\vec{p}_{\ell}(x) = \sum_{j} W_{\ell} e_{j} \otimes p_{j}(x) \in \mathbb{C}^{d \times v} \langle x \rangle_{k}$ , then (5.6) is further equivalent to

$$0 > \sum_{\ell} (\vec{p}_{\ell}(X)\gamma)^* \mathfrak{L}(X, Y) (\vec{p}_{\ell}(X)\gamma) = \lambda(q),$$
(5.7)

where  $q = \sum_{\ell} \vec{p}_{\ell}(x)^* \mathfrak{L}(x, y) \vec{p}_{\ell}(x)$  is a matrix polynomial only in *x* by (5.4), and thus  $q \in M_x^{\mu}(\mathfrak{L})_k$ . But now (5.7) contradicts the nonnegativity of  $\lambda$  on  $M_x^{\mu}(\mathfrak{L})_k$ .

The converse is obvious.

Proof of Theorem 5.1. Let  $\operatorname{Sym} \mathbb{C}^{\mu \times \mu} \langle x \rangle_{2r+1}$  denote the symmetric elements of  $\mathbb{C}^{\mu \times \mu} \langle x \rangle_{2r+1}$ . Arguing towards a contradiction, suppose  $p \in \operatorname{Sym} \mathbb{C}^{\mu \times \mu} \langle x \rangle_{2r+1}$  and  $p|_{\mathcal{K}} \geq 0$ , but  $p \notin M_x^{\mu}(\mathfrak{L})_r$ . By the scalar Hahn–Banach theorem and Proposition 5.3, there is a strictly separating positive (real) linear functional  $\lambda : \operatorname{Sym} \mathbb{C}^{\mu \times \mu} \langle x \rangle_{2r+1} \to \mathbb{R}$  nonnegative on  $M_x^{\mu}(\mathfrak{L})_r$ . We first extend  $\lambda$  to a (complex) linear functional on the whole

 $\mathbb{C}^{\mu \times \mu} \langle x \rangle_{2r+1}$  by sending q + is to  $\lambda(q) + i\lambda(s)$  for symmetric q, s. We then extend  $\lambda$  to a linear functional (still called  $\lambda$ ) on  $\mathbb{C}^{\mu \times \mu} \langle x \rangle_{2r+2}$  by

$$E_{i,j} \otimes u^* v \mapsto \begin{cases} 0 & \text{if } i \neq j \text{ or } u \neq v \\ C & \text{otherwise,} \end{cases}$$

where  $i, j = 1, ..., \mu$ , and  $u, v \in \langle x \rangle$  are of length r + 1. For C > 0 large enough, this  $\lambda$  will be nonnegative on  $\Sigma_{r+1}^{\mu}$ . Perturbing  $\lambda$  if necessary, we may further assume  $\lambda$  is strictly positive on  $\Sigma_r^{\mu} \setminus \{0\}$ . Now applying Proposition 5.4 yields a matrix tuple  $X \in \overline{\mathcal{K}}$  and a vector  $\gamma$  satisfying (5.3) (with k = r). But then

$$0 > \lambda(p) = \langle p(X)\gamma, \gamma \rangle \ge 0,$$

a contradiction.

#### 6. Tracial sets

While this paper's original motivation arose from considerations of free optimization as it appears in linear systems theory, determining the matrix convex hull of a free set has an analog in quantum information theory [LP11]. In free optimization, the relevant maps are completely positive and *unital* (ucp). In quantum information theory, the relevant maps are completely positive and *trace preserving* (CPTP) or *trace nonincreasing*. This section begins by recalling the two *quantum interpolation problems* from Subsection 3.2 before reformulating these problems in terms of *tracial hulls*. Corresponding duality results are the topic of the next section.

Recall a quantum channel is a cp map  $\Phi$  from  $M_n$  to  $M_k$  that is trace preserving,

$$\operatorname{tr}(\Phi(X)) = \operatorname{tr}(X).$$

The dual  $\Phi'$  of  $\Phi$  is the mapping from  $M_k$  to  $M_n$  defined by

$$\operatorname{tr}(\Phi(X)Y^*) = \operatorname{tr}(X\Phi'(Y)^*).$$

**Lemma 6.1** ([LP11, Proposition 1.2]).  $\Phi'$  *is a quantum channel cp if and only if*  $\Phi$  *is unital cp.* 

Recall the cp interpolation problem from Subsection 3.2. It asks, given  $A \in \mathbb{S}_n^g$  and  $B \in \mathbb{S}_m^g$ : does there exist a unital cp map  $\Phi : M_n \to M_m$  such that  $B_j = \Phi(A_j)$  for j = 1, ..., g? The set of solutions *B* for a given *A* is the matrix convex hull of *A*. The versions of the interpolation problem arising in quantum information theory [Hal1, Kle07, NCSB98] replace unital with trace preserving or trace nonincreasing. Namely, does  $B_j$  equal  $\Phi(A_j)$  for j = 1, ..., g for some trace preserving (resp. trace nonincreasing) cp map  $\Phi : M_n \to M_m$ ? The set of all solutions *B* for a given *A* is the *tracial hull* of *A*. Thus,

thull(A) = {B : 
$$\Phi(A) = B$$
 for some trace preserving cp map  $\Phi$ }. (6.1)

We define the *contractive tracial hull* of a tuple A by

 $\operatorname{cthull}(A) = \{B : \Phi(A) = B \text{ for some cp trace nonincreasing map } \Phi\}.$ 

The article [LP11] determines when  $B \in \text{thull}(A)$  for g = 1 (see Section 3.2). For any  $g \ge 0$  the paper [AG15, Section 3] converts this problem to an LMI suitable for semidefinite programming; see Theorem 3.4 here for a similar result.

While the unital and trace preserving (or trace nonincreasing) interpolation problems have very similar formulations, tracial hulls possess far less structure than matrix convex hulls. Indeed, as is easily seen, tracial hulls need not be convex (levelwise), and contractive tracial hulls need not be closed with respect to direct sums. Tracial hulls are studied in Subsection 6.1, and contractive tracial hulls in Subsection 6.2. Section 7 contains "tracial" notions of half-space and corresponding Hahn–Banach type separation theorems.

#### 6.1. Tracial sets and hulls

A set  $\mathcal{Y} \subseteq \mathbb{S}^g$  is *tracial* if whenever  $Y \in \mathcal{Y}(n)$  and  $C_\ell$  are  $m \times n$  matrices such that

$$\sum C_{\ell}^* C_{\ell} = I_n,$$

then  $\sum C_j Y C_j^* \in \mathcal{Y}(m)$ . The *tracial hull* of a subset  $S \subseteq \mathbb{S}^g$  is the smallest tracial set containing S, denoted thull(S). Note that if S is a singleton, this definition is consistent with the definition afforded by (6.1).

The following lemma is an easy consequence of a theorem of Choi, stated in [Pau02, Proposition 4.7]. It caps the number of terms needed in a convex combination to represent a given matrix tuple Z in the tracial hull of T. Hence it is an analog of Carathéodory's convex hull theorem (see e.g. [Bar02, Theorem I.2.3]).

**Lemma 6.2.** Suppose  $T \in \mathbb{S}_n^g$  and  $C_1, \ldots, C_N$  are  $m \times n$  matrices with  $\sum C_\ell^* C_\ell = I_n$ . If  $Z = \sum_{\ell=1}^N C_\ell T C_\ell^*$ , then there exist  $m \times n$  matrices  $V_1, \ldots, V_{mn}$  such that  $\sum V_\ell^* V_\ell = I_n$  and

$$Z = \sum_{\ell=1}^{mn} V_{\ell} T V_{\ell}^*.$$

*Proof.* The mapping  $\Phi: M_n \to M_m$  defined by

$$\Phi(X) = \sum C_{\ell} X C_{\ell}^*$$

is completely positive. Hence, by [Pau02, Proposition 4.7], there exist (at most) nm matrices  $V_j : \mathbb{C}^m \to \mathbb{C}^n$  such that

$$\Phi(X) = \sum_{\ell=1}^{mn} V_\ell X V_\ell^*.$$

In particular,

$$Z = \Phi(T) = \sum V_{\ell} T V_{\ell}^*$$

Further, for all  $m \times m$  matrices X,

$$\operatorname{tr}(X) = \operatorname{tr}\left(X \sum C_{\ell}^* C_{\ell}\right) = \operatorname{tr}\left(\sum C_{\ell} X C_{\ell}^*\right) = \operatorname{tr}(\Phi(X))$$
$$= \operatorname{tr}\left(\sum V_{\ell} X V_{\ell}^*\right) = \operatorname{tr}\left(X \sum V_{\ell}^* V_{\ell}\right).$$

It follows that  $\sum V_{\ell}^* V_{\ell} = I$ .

**Lemma 6.3.** For  $S = \{T\}$  a singleton,

$$\operatorname{thull}(\{T\}) = \left\{ \sum C_{\ell} T C_{\ell}^* : \sum C_{\ell}^* C_{\ell} = I \right\}$$

Moreover, this set is closed (levelwise).

The tracial hull of a subset  $S \subseteq \mathbb{S}^g$  is

thull(
$$\mathcal{S}$$
) =  $\left\{ \sum C_{\ell} T C_{\ell}^* : \sum C_{\ell}^* C_{\ell} = I, T \in \mathcal{S} \right\} = \bigcup_{T \in \mathcal{S}} \text{thull}(\{T\}).$ 

If S is a finite set, then the tracial hull of S is closed.

*Proof.* The first statement follows from the observation that  $\{\sum C_{\ell} T C_{\ell}^* : \sum C_{\ell}^* C_{\ell} = I\}$  is tracial.

To prove the "moreover" part, suppose T has size n and suppose  $Z^k$  is a sequence from  $\mathcal{Y}(m)$ . By Lemma 6.2 for each k there exist nm matrices  $V_{k,\ell}$  of size  $n \times m$  such that

$$Z^k = \sum_{\ell} V_{k,\ell} T V_{k,\ell}^*$$

and each  $V_{k,\ell}$  is a contraction. Hence, by passing to a subsequence, we can assume that for each fixed  $\ell$ , the sequence  $(V_{k,\ell})_k$  converges to some  $W_\ell$ . Hence  $Z^k$  converges to  $Z = \sum_{\ell} W_{\ell} T W_{\ell}^*$ . Also, since  $\sum_{\ell} V_{k,\ell}^* V_{k,\ell} = I$  for each k, we have  $\sum_{\ell} W_{\ell}^* W_{\ell} = I$ , whence  $Z \in \mathcal{Y}(m)$ .

To prove the second statement, let  $S \subseteq \mathbb{S}^{g}$ . Evidently,

$$S \subseteq \bigcup_{T \in S} \operatorname{thull}(\{T\}) \subseteq \operatorname{thull}(S).$$

Hence it suffices to show that  $\bigcup_{T \in S} \text{thull}(\{T\})$  is itself tracially convex. To this end, suppose  $X \in \bigcup_{T \in S} \text{thull}(\{T\})$  and  $C_1, \ldots, C_N$  with  $\sum C_{\ell}^* C_{\ell} = I$  are given (and of the appropriate sizes). There is an  $S \in S$  such that  $X \in \text{thull}(\{S\})$ . Hence, by the first part of the lemma,  $\sum C_{\ell} X C_{\ell}^* \in \text{thull}(\{S\}) \subseteq \bigcup_{T \in S} \text{thull}(\{T\})$ , and the desired conclusion follows.

The final statement of the lemma follows by combining its first two assertions and using the fact that the closure of a finite union is the finite union of the closures.  $\Box$ 

#### 6.2. Contractively tracial sets and hulls

A set  $\mathcal{Y} \subseteq \mathbb{S}^g$  is *contractively tracial* if whenever  $Y \in \mathcal{Y}(m)$  and  $C_\ell$  are  $n \times m$  matrices such that

$$\sum C_{\ell}^* C_{\ell} \preceq I_m, \tag{6.2}$$

then  $\sum C_j Y C_j^* \in \mathcal{Y}(n)$ . Note that, in this case,  $\mathcal{Y}$  is closed under unitary conjugation and compression to subspaces, but not necessarily direct sums. It is clear that intersections of contractively tracial sets are again contractively tracial.

If S is a singleton, the *contractive tracial hull* of a set S, defined as the smallest contractively tracial set containing S, is consistent with our earlier definition in terms of cp maps.

**Lemma 6.4.** The contractive tracial hull of a subset  $S \subseteq \mathbb{S}^g$  is

$$\operatorname{cthull}(\mathcal{S}) = \left\{ \sum C_{\ell} T C_{\ell}^* : \sum C_{\ell}^* C_{\ell} \leq I, \ T \in \mathcal{S} \right\} = \bigcup_{T \in \mathcal{S}} \operatorname{cthull}(\{T\}).$$

If S is a finite set, then the contractive tracial hull of S is closed.

*Proof.* The proof is the same as for Lemma 6.3, so is omitted.

Tracial and contractively tracial sets are not necessarily convex, as Example 8.6 illustrates, and they are not necessarily free sets because they may not respect direct sums. Lemma 6.5 below explains the relation between these two failings. Recall that a subset  $\mathcal{Y}$  of  $\mathbb{S}^g$  is levelwise convex if each  $\mathcal{Y}(n)$  is convex (in the usual sense as a subset of  $\mathbb{S}_n^g$ ). Say that  $\mathcal{Y}$  is *closed with respect to convex direct sums* if given  $\ell$  and  $Y^1, \ldots, Y^{\ell} \in \mathcal{Y}$  and given  $\lambda_1, \ldots, \lambda_{\ell} \geq 0$  with  $\sum \lambda_j \leq 1$ , we have

$$\bigoplus_j \lambda_j Y^j \in \mathcal{Y}.$$

**Lemma 6.5.** If  $\mathcal{Y}$  is contractively tracial, then  $\mathcal{Y}$  is levelwise convex if and only if  $\mathcal{Y}$  is closed with respect to convex direct sums.

*Proof.* Suppose each  $\mathcal{Y}(m)$  is convex. Given  $Y^j \in \mathcal{Y}(m_j)$  for  $1 \leq j \leq \ell$ , let  $m = \sum m_j$ . Consider the block operator column  $W_j$  embedding  $\mathbb{C}^{m_j}$  into  $\mathbb{C}^m = \bigoplus_j \mathbb{C}^{m_j}$ . Note that  $W_j^* W_j = I_{m_j}$  and thus contractively tracial implies  $W_j Y^j W_j^* \in \mathcal{Y}(m)$ . Hence, given  $\lambda_j \geq 0$  with  $\sum \lambda_j = 1$ , convexity of  $\mathcal{Y}(m)$  (in the ordinary sense) implies

$$\bigoplus_{j} \lambda_{j} Y^{j} = \sum \lambda_{j} W_{j} Y^{j} W_{j}^{*} \in \mathcal{Y}(m).$$

To prove the converse, suppose  $Y^j \in \mathcal{Y}(n)$  and  $m = \ell n$ . In this case,  $\sum W_j W_j^* = I_n$ , and hence tracial implies

$$\sum W_j^* \left( \bigoplus \lambda_j Y^j \right) W_j = \sum \lambda_j Y^j \in \mathcal{Y}(n).$$

# 6.3. Classical duals of free convex hulls and of tracial hulls

This subsection gives properties of the classical polar dual of matrix convex hulls and tracial hulls. Real linear functionals  $\lambda : \mathbb{S}_n^g \to \mathbb{R}$  are in one-one correspondence with elements  $B \in \mathbb{S}_n^g$  via the pairing

$$\lambda(X) = \operatorname{tr}\left(\sum B_j X_j\right), \quad X = (X_1, \dots, X_g).$$

Write  $\lambda_B$  for this  $\lambda$ . To avoid confusion with the free polar duals appearing earlier in this article, let  $\mathcal{U}^{\circ c}$  denote the *conventional polar dual* of a subset  $\mathcal{U} \subseteq \mathbb{S}_n^g$ ,

$$\mathcal{U}^{\circ c} = \{ B \in \mathbb{S}_n^g : \lambda_B(X) \le 1 \text{ for all } X \in \mathcal{U} \}.$$

~

**Lemma 6.6.** Suppose  $A \in \mathbb{S}_n^g$ .

- (i)  $\operatorname{co}^{\operatorname{mat}}(A)^{\circ c} = \{Y : \operatorname{thull}(Y) \subseteq \{A\}^{\circ c}\};$
- (ii) thull  $(A)^{\circ c} = \{Y : \{A\}^{\circ c} \supseteq \operatorname{co}^{\operatorname{mat}}(Y)\};$  and

(iii) thull(B)  $\subseteq$  thull(A) if and only if  $\{A\}^{\circ c} \supseteq \operatorname{co}^{\operatorname{mat}}(Y)$  implies  $\{B\}^{\circ c} \supseteq \operatorname{co}^{\operatorname{mat}}(Y)$ .

*Proof.* The first formula:

$$\operatorname{co}^{\operatorname{mat}}(A)^{\circ c} = \left\{ Y : 1 - \operatorname{tr}\left(\sum_{j} V_{j}^{*} A V_{j} Y\right) \ge 0, \sum_{j} V_{j}^{*} V_{j} = I \right\}$$
$$= \left\{ Y : 1 - \operatorname{tr}\left(A \sum_{j} V_{j} Y V_{j}^{*}\right) \ge 0, \sum_{j} V_{j}^{*} V_{j} = I \right\}$$
$$= \left\{ Y : 1 - \operatorname{tr}(AG) \ge 0, G \in \operatorname{thull}(Y) \right\}$$
$$= \left\{ Y : \left\{A\right\}^{\circ c} \supseteq \operatorname{thull}(Y) \right\}.$$

The second formula:

$$\operatorname{thull}(A)^{\circ c} = \left\{ Y : 1 - \operatorname{tr}\left(\sum_{j} V_{j}^{*} A V_{j} Y\right) \ge 0, \sum_{j} V_{j} V_{j}^{*} = I \right\}$$
$$= \left\{ Y : 1 - \operatorname{tr}\left(A \sum_{j} V_{j} Y V_{j}^{*}\right) \ge 0, \sum_{j} V_{j} V_{j}^{*} = I \right\}$$
$$= \left\{ Y : \{A\}^{\circ c} \supseteq \operatorname{co}^{\operatorname{mat}}(Y) \right\}.$$

The third formula: thull(*B*)  $\subseteq$  thull(*A*) if and only if thull(*B*)<sup>oc</sup>  $\supseteq$  thull(*A*)<sup>oc</sup>, if and only if

$$\left\{Y: \{B\}^{\circ c} \supseteq \operatorname{co}^{\mathrm{mat}}(Y)\right\} \supseteq \left\{Y: \{A\}^{\circ c} \supseteq \operatorname{co}^{\mathrm{mat}}(Y)\right\},\$$

if and only if  $\{A\}^{\circ c} \supseteq \operatorname{co}^{\operatorname{mat}}(Y)$  and  $\{B\}^{\circ c} \supseteq \operatorname{co}^{\operatorname{mat}}(Y)$ .

# 7. Tracial spectrahedra and an Effros-Winkler separation theorem

Classically, convex sets are delineated by half-spaces. In this section a notion of half-space suitable in the tracial context—we call these *tracial spectrahedra*—is introduced. Subsection 7.3 contains a free Hahn–Banach separation theorem for tracial spectrahedra. The section concludes with applications of this Hahn–Banach theorem. Subsection 7.4 suggests several notions of duality based on the tracial separation theorem from Subsection 7.3. Subsection 7.5 studies free (convex) cones.

#### 7.1. Tracial spectrahedra

Polar duality considerations in the trace nonincreasing context lead naturally to inequalities of the type

$$I\otimes T-\sum_{j=1}^g B_j\otimes Y_j\succeq 0$$

for tuples  $B, Y \in \mathbb{S}^g$  and a positive semidefinite matrix T with trace at most one. Two notions, in a sense dual to one another, of half-space are obtained by fixing either B or Y.

Given  $B \in \mathbb{S}_k^g$ , let

$$\begin{split} \tilde{\mathfrak{H}}_B &= \bigcup_{m \in \mathbb{N}} \Big\{ Y \in \mathbb{S}_m^g : \exists T \succeq 0, \ \mathrm{tr}(T) \le 1, \ I \otimes T - \sum B_j \otimes Y_j \succeq 0 \Big\} \\ &= \bigcup_{m \in \mathbb{N}} \Big\{ Y \in \mathbb{S}_m^g : \exists T \succeq 0, \ \mathrm{tr}(T) = 1, \ I \otimes T - \sum B_j \otimes Y_j \succeq 0 \Big\}. \end{split}$$

We call sets of the form  $\mathfrak{H}_B$  *tracial spectrahedra*. Tracial spectrahedra obtained by fixing *Y*, and parametrizing over *B*, appear in Subsubsection 7.4.2.

**Proposition 7.1.** Let  $B \in \mathbb{S}_k^g$ . Then

- (a) the set  $\mathfrak{H}_B$  is contractively tracial;
- (b) for each m, the set  $\mathfrak{H}_B(m)$  is convex; and

(c) for each m, the set  $\mathfrak{H}_B(m)$  is closed.

In summary,  $\mathfrak{H}_B$  is levelwise compact and closed, and it is contractively tracial.

**Remark 7.2.** Of course  $\mathfrak{H}_B$  is not a free set since, in particular, it is not closed with respect to direct sums.

*Proof of Proposition 7.1.* Suppose  $Y \in \mathfrak{H}_B(m)$  and  $C_\ell$  satisfy (6.2). There is an  $m \times m$  positive semidefinite matrix T with trace at most one such that

$$I\otimes T-\sum B_j\otimes Y_j\succeq 0.$$

It follows that

$$0 \leq I \otimes \sum_{\ell} C_{\ell} T C_{\ell}^* - \sum_{j} B_j \otimes \sum_{\ell} C_{\ell} Y_j C_{\ell}^*.$$

Note that  $T' = \sum_{\ell} C_{\ell} T C_{\ell}^* \succeq 0$  and

$$\operatorname{tr}(T') = \operatorname{tr}\left(T \sum C_{\ell}^* C_{\ell}\right) = \operatorname{tr}(T^{1/2} C_{\ell}^* C_{\ell} T^{1/2}) \le \operatorname{tr}(T) \le 1.$$

Hence  $\sum C_{\ell} Y C_{\ell}^* \in \mathcal{Y}(n)$ , and (a) is proved.

To prove (b), suppose both  $Y^1$  and  $Y^2$  are in  $\mathcal{Y}_B$ . To each there is an associated positive semidefinite matrix of trace at most one, say  $T_1$  and  $T_2$ . If  $0 \le s_1, s_2 \le 1$  and  $s_1 + s_2 = 1$ , then  $T = \sum_{\ell} s_{\ell} T_{\ell}$  is positive semidefinite and has trace at most one. Moreover, with  $Y = \sum s_j Y^j$ ,

$$I \otimes T - \sum_{j} B_{j} \otimes \left( \sum s_{\ell} Y_{j}^{\ell} \right) = \sum_{\ell} s_{\ell} \left( I \otimes T_{\ell} - \sum_{j} B_{j} \otimes Y_{j}^{\ell} \right) \succeq 0.$$

To prove (c), suppose the sequence  $(Y^k)_k$  from  $\mathfrak{H}_B(m)$  converges to  $Y \in \mathbb{S}_m^g$ . For each k there is a positive semidefinite matrix  $T_k$  of trace at most one such that

$$I \otimes T_k - \Lambda_B(Y^k) \succeq 0.$$

Choose a convergent subsequence of the  $T_k$  with limit T. This T witnesses  $Y \in \mathfrak{H}_B(m)$ .

To proceed toward the separation theorem we start with some preliminaries.

## 7.2. An auxiliary result

Given a positive integer *n*, let  $\mathcal{T}_n$  denote the positive semidefinite  $n \times n$  matrices of trace one. Each  $T \in \mathcal{T}_n$  corresponds to a state on  $M_n$  via the trace

$$M_n \ni A \mapsto \operatorname{tr}(AT). \tag{7.1}$$

Conversely, to each state  $\varphi$  on  $M_n$  we can assign a matrix T such that  $\varphi$  is the map (7.1). Note that  $\mathcal{T}_n$  is a convex, compact subset of  $\mathbb{S}_n$ , the symmetric  $n \times n$  matrices.

The following lemma is a version of [EW97, Lemma 5.2]. An affine (real) linear mapping  $f : \mathbb{S}_n \to \mathbb{R}$  is a function of the form  $f(x) = a_f + \lambda_f(x)$ , where  $\lambda_f$  is (real) linear and  $a_f \in \mathbb{R}$ .

**Lemma 7.3.** Suppose  $\mathcal{F}$  is a convex set of affine linear mappings  $f : \mathbb{S}_n \to \mathbb{R}$ . If for each  $f \in \mathcal{F}$  there is a  $T \in \mathcal{T}_n$  such that  $f(T) \ge 0$ , then there is a  $\mathfrak{T} \in \mathcal{T}_n$  such that  $f(\mathfrak{T}) \ge 0$  for every  $f \in \mathcal{F}$ .

*Proof.* For  $f \in \mathcal{F}$ , let

$$B_f = \{T \in \mathcal{T}_n : f(T) \ge 0\} \subseteq \mathcal{T}_n.$$

By hypothesis each  $B_f$  is nonempty and it suffices to prove that

$$\bigcap_{f\in\mathcal{F}}B_f\neq\emptyset.$$
Since each  $B_f$  is compact, it suffices to prove that the collection  $\{B_f : f \in \mathcal{F}\}\$  has the finite intersection property. Accordingly, let  $f_1, \ldots, f_m \in \mathcal{F}$  and suppose

$$\bigcap_{j=1}^m B_{f_j} = \emptyset.$$

Define  $F : \mathbb{S}_n \to \mathbb{R}^m$  by

$$F(T) = (f_1(T), \ldots, f_m(T)).$$

Then  $F(\mathcal{T}_n)$  is both convex and compact because  $\mathcal{T}_n$  is both convex and compact and each  $f_i$ , and hence F, is affine linear. Moreover,  $F(\mathcal{T}_n)$  does not intersect

$$\mathbb{R}_{>0}^{m} = \{x = (x_1, \dots, x_m) \in \mathbb{R}^m : x_j \ge 0 \text{ for each } j\}$$

Hence there is a linear functional  $\lambda : \mathbb{R}^m \to \mathbb{R}$  such that

$$\lambda(F(\mathcal{T}_n)) < 0 \text{ and } \lambda(\mathbb{R}^m_{>0}) \ge 0.$$

There exists  $\lambda_j \in \mathbb{R}$  such that  $\lambda(x) = \sum \lambda_j x_j$ . Since  $\lambda(\mathbb{R}^m_{\geq 0}) \geq 0$  it follows that  $\lambda_j \geq 0$  for all *j*, and since  $\lambda \neq 0$ , for at least one *k* we have  $\lambda_k > 0$ . Without loss of generality, it may be assumed that  $\sum \lambda_j = 1$ . Let

$$f=\sum \lambda_j f_j.$$

Since  $\mathcal{F}$  is convex, it follows that  $f \in \mathcal{F}$ . On the other hand,  $f(T) = \lambda(F(T))$ . Hence if  $T \in \mathcal{T}_n$ , then f(T) < 0. Thus, for this f there does not exist a  $T \in \mathcal{T}_n$  such that  $f(T) \ge 0$ , a contradiction which completes the proof.

## 7.3. A tracial spectrahedron separating theorem

The following lemma is proved by a variant of the Effros–Winkler construction of separating LMIs (i.e., the matricial Hahn–Banach theorem) in the theory of matrix convex sets.

**Lemma 7.4.** Fix positive integers m, n, and suppose that S is a nonempty subset of  $\mathbb{S}_m^g$ . Let U denote the subset of  $\mathbb{S}_n^g$  consisting of all tuples of the form

$$\sum_{\ell=1}^{\mu} C_{\ell} Y^{\ell} C_{\ell}^*,$$

where each  $C_{\ell}$  is  $n \times m$ , each  $Y^{\ell}$  is in S and  $\sum C_{\ell}^* C_{\ell} \leq I$ . If  $B \in \mathbb{S}_n^g$  is in the conventional polar dual of U, then there exists a positive semidefinite  $m \times m$  matrix T with trace at most one such that

$$I \otimes T - \sum B_j \otimes Y_j \succeq 0$$
 for every  $Y \in S$ .

*Proof.* Recall the definition of  $\lambda_B$  from Subsection 6.3. Given  $C_\ell$  and  $Y^\ell$  as in the statement of the lemma, define  $f_{C,Y} : \mathbb{S}_m^g \to \mathbb{R}$  by

$$f_{C,Y}(X) = \operatorname{tr}\left(\sum C_{\ell} X C_{\ell}^*\right) - \lambda_B\left(\sum C_{\ell} Y C_{\ell}^*\right).$$

Let  $\mathcal{F} = \{f_{C,Y} : C, Y\}$ . Thus  $\mathcal{F}$  is a set of affine (real) linear mappings from  $\mathbb{S}_m^g$  to  $\mathbb{R}$ . To show that  $\mathcal{F}$  is convex, suppose, for  $1 \le s \le N$ ,  $C^s = (C_1^s, \ldots, C_{\mu_s}^s)$  is a tuple of  $n \times m$  matrices, for  $1 \le s \le N$  and  $1 \le j \le \mu_s$  the matrices  $Y^{s,j}$  are in  $\mathcal{S}$ , and  $\lambda_1, \ldots, \lambda_N$  are positive numbers with  $\sum \lambda_s = 1$ . In this case,

$$\sum \lambda_s f_{C^s, Y^{s, \cdot}} = f_{C, Y}$$

for

$$C = \left(\frac{1}{\sqrt{\lambda_s}}C_\ell^s\right)_{s,\ell}, \quad Y = (Y^{s,\ell})_{s,\ell}.$$

Hence  $\mathcal{F}$  is convex.

Given  $n \times m$  matrices  $C_1, \ldots, C_{\mu}$  and  $Y^1, \ldots, Y^{\mu} \in S$ , let  $D = \sum C_{\ell}^* C_{\ell}$ . Assuming D has norm one, there is a unit vector  $\gamma$  such that  $||D\gamma|| = ||D|| = 1$ . Choose  $T = \gamma \gamma^*$ . Thus  $T \in \mathcal{T}_m$ . Moreover,

$$\operatorname{tr}\left(\sum C_{\ell}TC_{\ell}^{*}\right) = \operatorname{tr}(TD) = \langle D\gamma, \gamma \rangle = 1.$$

Thus, using the assumption that *B* is in  $\mathcal{U}^{\circ c}$ , we see that

$$f_{C,Y}(T) = 1 - \lambda_B \left( \sum C_\ell Y^\ell C_\ell^* \right) \ge 0.$$

If D is not of norm one, a simple scaling argument gives the same conclusion,

$$f_{C,Y}(T) \ge 0.$$

Thus, for each  $f \in \mathcal{F}$  there exists a  $T \in \mathcal{T}_m$  such that  $f(T) \ge 0$ . By Lemma 7.3, it follows that there is a  $\mathfrak{T} \in \mathcal{T}_m$  such that  $f_C(\mathfrak{T}) \ge 0$  for all C and Y, i.e.,

$$\operatorname{tr}\left(\sum C_{\ell} \mathfrak{T} C_{\ell}^{*}\right) - \lambda_{B}\left(\sum C_{\ell} Y^{\ell} C_{\ell}^{*}\right) \geq 0, \qquad (7.2)$$

regardless of the norm of  $\sum C_{\ell}^* C_{\ell}$ .

Now the aim is to show that

$$\Delta := I \otimes \mathfrak{T} - \sum_{j} B_{j} \otimes Y_{j} \succeq 0 \quad \text{for every } Y \in \mathcal{S}.$$

Accordingly, let  $Y \in S$  and  $\gamma = \sum e_s \otimes \gamma_s \in \mathbb{R}^n \otimes \mathbb{R}^m$ . Compute

$$\langle \Delta \gamma, \gamma \rangle = \sum_{s} \langle \mathfrak{T} \gamma_{s}, \gamma_{s} \rangle - \sum_{j} \sum_{s,t} (B_{j})_{s,t} \langle Y_{j} \gamma_{s}, \gamma_{t} \rangle$$

Now let  $\Gamma^*$  denote the matrix with *s*-th column  $\gamma_s$ . Hence  $\Gamma$  is  $n \times m$  and

$$\lambda_B(\Gamma Y \Gamma^*) = \operatorname{tr}\left(\sum B_j(\Gamma Y_j \Gamma^*)\right) = \sum_j \sum_{s,t} (B_j)_{s,t} \langle Y_j \gamma_s, \gamma_t \rangle.$$

Similarly,

$$\operatorname{tr}(\Gamma\mathfrak{T}\Gamma^*)=\sum_{s}\langle\mathfrak{T}\gamma_s,\gamma_s\rangle.$$

Thus, by (7.2),

$$\langle \Delta \gamma, \gamma \rangle = \operatorname{tr}(\Gamma \mathfrak{T} \Gamma^*) - \lambda_B(\Gamma Y \Gamma^*) \ge 0$$

It is in this last step that the contractively tracial, not just tracial is needed, so that it is not necessary for  $\Gamma^*\Gamma$  to be a multiple of the identity.

**Proposition 7.5.** If  $\mathcal{Y} \subseteq \mathbb{S}^g$  is contractively tracial and if  $B \in \mathbb{S}_n^g$  is in the conventional polar dual  $\mathcal{Y}(n)^{\circ c}$  of  $\mathcal{Y}(n)$ , then  $\mathcal{Y} \subseteq \mathfrak{H}_B$ .

*Proof.* Suppose  $\mathcal{Y}$  is contractively tracial and  $Y \in \mathcal{Y}(m)$ . Letting  $\mathcal{S} = \{Y\}$  in Lemma 7.4, it follows that there is a *T* such that

$$I\otimes T-\sum B_j\otimes Y_j\succeq 0.$$

Thus,  $Y \in \mathfrak{H}_B$ , and the proof is complete.

We are now ready to state the separation result for closed levelwise convex tracial sets.

**Theorem 7.6.** (i) If  $\mathcal{Y} \subseteq \mathbb{S}^g$  is contractively tracial, levelwise convex, and if  $Z \in \mathbb{S}_m^g$  is not in the closure of  $\mathcal{Y}(m)$ , then there exists a  $B \in \mathbb{S}_m^g$  such that  $\mathcal{Y} \subseteq \mathfrak{H}_B$ , but  $Z \notin \mathfrak{H}_B$ . Hence,

$$\overline{\mathcal{Y}} = \bigcap \{\mathfrak{H}_B : \mathfrak{H}_B \supseteq \mathcal{Y}\} = \bigcap_{n \in \mathbb{N}} \bigcap_{B \in \mathcal{Y}(n)^{\circ c}} \mathfrak{H}_B.$$

(ii) The levelwise closed convex contractively tracial hull of a subset  $\mathcal{Y}$  of  $\mathbb{S}^g$  is

$$\bigcap \{\mathfrak{H}_B : \mathfrak{H}_B \supseteq \mathcal{Y}\}.$$

*Proof.* To prove (i), suppose  $Z \in \mathbb{S}_m^g$  but  $Z \notin \overline{\mathcal{Y}(m)}$ . Since  $\mathcal{Y}$  is levelwise convex, there is  $\lambda_B$  such that  $\lambda_B(Y) \leq 1$  for all  $Y \in \mathcal{Y}(m)$ , but  $\lambda_B(Z) > 1$  by the usual Hahn–Banach separation theorem for closed convex sets. Thus *B* is in the conventional polar dual of  $\mathcal{Y}(m)^{\circ c}$ . From Proposition 7.5,  $\mathcal{Y} \subseteq \mathfrak{H}_B$ .

On the other hand, if  $T \in \mathcal{T}_m$  and  $\{e_1, \ldots, e_m\}$  is an orthonormal basis for  $\mathbb{R}^m$ , then, with  $e = \sum e_s \otimes e_s \in \mathbb{R}^m \otimes \mathbb{R}^m$ ,

$$\left\langle \left(I \otimes T - \sum B_j \otimes Z_j\right)e, e \right\rangle = \operatorname{tr}(T) - \operatorname{tr}\left(\sum B_j Z_j\right) = 1 - \lambda_B(Z) < 0$$

Hence  $Z \notin \mathfrak{H}_B$  and the conclusion follows.

To prove (ii), first note that if  $\mathcal{I}$  denotes the intersection of the  $\mathfrak{H}_B$  that contain  $\mathcal{Y}$ , then  $\mathcal{Y} \subseteq \mathcal{I}$ . Since the intersection of tracial spectrahedra is levelwise closed and convex, and contractively tracial, the levelwise closed convex tracial hull  $\mathcal{H}$  of  $\mathcal{Y}$  is also contained in  $\mathcal{I}$ . On the other hand, from (i),

$$\mathcal{H} = \bigcap \{ \mathfrak{H}_B : \mathfrak{H}_B \supseteq \mathcal{H} \} \supseteq \mathcal{I} \supseteq \mathcal{H}.$$

**Remark 7.7** (The contractive tracial hull of a point). Fix a  $Y \in \mathbb{S}_n^g$  and let  $\mathcal{Y}$  denote its contractive tracial hull,

$$\mathcal{Y} = \left\{ \sum V_j Y V_j^* : \sum V_j^* V_j \preceq I \right\}.$$

Evidently each  $\mathcal{Y}(m)$  (taking  $V_j : \mathbb{R}^n \to \mathbb{R}^m$ ) is a convex set. From Lemma 6.4,  $\mathcal{Y}$  is closed. Hence Theorem 7.6 applies and gives a duality description of  $\mathcal{Y}$ . Namely,  $\tilde{Y}$  is in the contractive tracial hull  $\mathcal{Y}$  if and only if for each B for which there exists a positive semidefinite T of trace at most one such that

$$I\otimes T-\sum B_j\otimes Y_j\succeq 0$$

there exists a positive semidefinite  $\tilde{T}$  of trace at most one such that

$$I\otimes \tilde{T}-\sum B_j\otimes \tilde{Y}_j\succeq 0.$$

# 7.4. Tracial polar duals

We now introduce two natural notions of polar duals based on the tracial spectrahedra. Rather than exhaustively studying these duals, we list a few properties to illustrate the possibilities.

7.4.1. Ex situ tracial dual. Suppose  $\mathcal{K} \subseteq \mathbb{S}^g$ . Let  $\hat{\mathcal{K}}$  denote its *ex situ tracial dual* defined by

$$\hat{\mathcal{K}} = \bigcap_{B \in \mathcal{K}} \mathfrak{H}_B.$$

Thus,

$$\hat{\mathcal{K}}(n) = \left\{ Y \in \mathbb{S}_n^g : \forall B \in \mathcal{K} \; \exists T \succeq 0, \; \mathrm{tr}(T) \le 1, \; I \otimes T - \sum B_j \otimes Y_j \succeq 0 \right\}.$$

**Proposition 7.8.** If  $\mathcal{K}$  is matrix convex and each  $\mathcal{K}^{\circ}(n)$  is bounded (equivalently,  $\mathcal{K}(1)$  contains 0 in its interior), then

(i)  $\hat{\mathcal{K}}(n) = \{Y \in \mathbb{S}_n^g : \exists T \succeq 0, \operatorname{tr}(T) \le 1 \forall B \in \mathcal{K}, I \otimes T - \sum B_j \otimes Y_j \succeq 0\};$ (ii)  $\hat{\mathcal{K}}(n) = \{SMS : M \in \mathcal{K}^\circ(n), S \succeq 0, \operatorname{tr}(S^2) \le 1\}.$ 

*Proof.* Suppose *K* is matrix convex. To prove (i), let  $Y \in \hat{\mathcal{K}}(n)$ . For each *B*, let  $\mathcal{T}_B = \{T \in \mathcal{T}_n : I \otimes T - \sum B_j \otimes Y_j \succeq 0\}$ . Thus, the hypothesis that  $Y \in \hat{\mathcal{K}}(n)$  is equivalent to assuming that for every *B* in  $\mathcal{K}$ , the set  $\mathcal{T}_B$  is nonempty.

That  $\mathcal{T}_B$  is compact will be verified by showing it has the finite intersection property. Now given  $B^1, \ldots, B^\ell \in \mathcal{K}$ , let  $B = \bigoplus_k B^k \in \mathcal{K}$ . Since  $B \in \mathcal{K}$ , there is a T such that

$$\bigoplus_k \left( I \otimes T - \sum B_j^k \otimes Y_j \right) = I \otimes T - \sum B_j \otimes Y_j \succeq 0.$$

Hence  $T \in \bigcap_{k=1}^{\ell} \mathcal{T}_{B^k}$ . It follows that the collection  $\{\mathcal{T}_B : B \in \mathcal{K}\}$  has the finite intersection property and hence there is a  $T \in \bigcap_{B \in \mathcal{K}} \mathcal{T}_B$ , and the forward inclusion in (i) follows. The reverse inclusion holds whether or not  $\mathcal{K}$  is matrix convex.

To prove (ii), suppose  $Y \in \hat{\mathcal{K}}(n)$ . Thus, by what has already been proved, there is a positive semidefinite matrix S such that  $tr(S^2) \le 1$  and

$$I \otimes S^2 - \sum B_j \otimes Y_j \succeq 0 \tag{7.3}$$

for all  $B \in \mathcal{K}$ . For positive integers k, let  $S_k^+$  denote the inverse of S + 1/k. Multiplying (7.3) on the left and on the right by  $I \otimes S_k^+$  yields

$$I \otimes P - \sum B_j \otimes S_k^+ Y_j S_k^+ \succeq 0$$

where *P* is the projection onto the range of *S*. It follows that  $M_k = S_k^+ Y S_k^+ \in \mathcal{K}^\circ(n)$ . Since  $\mathcal{K}^\circ(n)$  is bounded (by assumption) and closed, it is compact, and consequently a subsequence of  $(M_k)_k$  converges to some  $M \in \mathcal{K}^\circ(n)$ . Hence, Y = SMS.

Reversing the argument above shows that if  $M \in \mathcal{K}^{\circ}(n)$  and *S* is positive semidefinite with  $tr(S^2) \leq 1$ , then  $Y = SMS \in \hat{\mathcal{K}}(n)$ , and the proof is complete.

**Proposition 7.9.** The ex situ tracial dual  $\hat{\mathcal{K}}$  of a free spectrahedron  $\mathcal{K} = \mathcal{D}_{\mathfrak{L}_{\Omega}}$  is exactly the set

$$\Big\{\sum_{\ell} C_{\ell}^* \Omega C_{\ell} : \operatorname{tr}\Big(\sum C_{\ell}^* C_{\ell}\Big) \le 1\Big\}.$$

*Proof.* Suppose Y is in the ex situ tracial dual. By Proposition 7.8, there is a positive semidefinite matrix S with  $tr(S^2) \le 1$  and an  $M \in \mathcal{K}^\circ$  such that Y = SMS. Since  $M \in \mathcal{K}^\circ$ , by Remark 4.8 there is a positive integer  $\mu$  and a contraction V such that

$$M = V^*(I_\mu \otimes \Omega)V = \sum_{k=1}^{\mu} V_k^* \Omega V_k.$$

Hence,

$$Y = \sum_{k} S V_k^* \Omega V_k S.$$

Finally,

$$\operatorname{tr}\left(\sum SV_k^*V_kS\right) \le \operatorname{tr}(S^2) \le 1.$$

Conversely, suppose tr $(\sum C_{\ell}^* C_{\ell}) \leq 1$  and  $Y = \sum C_{\ell}^* \Omega C_{\ell}$ . Let  $T = \sum C_{\ell}^* C_{\ell}$  and note that for  $B \in \mathcal{K}$ ,

$$I \otimes T - \sum B_j \otimes Y_j = \sum_{\ell} C_{\ell}^* \Big( I \otimes I - \sum_j B_j \otimes \Omega_j \Big) C_{\ell} \succeq 0.$$

7.4.2. In situ tracial dual. Given a free set  $\mathcal{K} \subseteq \mathbb{S}^g$ , we can define another dual set we call the *in situ*  $\mathcal{K}^{\triangleright} = (\mathcal{K}^{\triangleright}(m))_m$  by

$$\mathcal{K}^{\triangleright}(m) = \{ B \in \mathbb{S}_m^g : \mathcal{K} \subseteq \mathfrak{H}_B \}.$$

Equivalently,

$$\mathcal{K}^{\triangleright}(m) = \Big\{ B \in \mathbb{S}_m^g : \forall Y \in \mathcal{K} \; \exists T \succeq 0, \; \mathrm{tr}(T) \le 1, \; I \otimes T - \sum B_j \otimes Y_j \succeq 0 \Big\}.$$

Each  $\mathcal{K}^{\triangleright}(m)$  is levelwise convex. Moreover, if  $B \in \mathcal{K}^{\triangleright}$  and  $V^*V \leq I$ , then  $V^*BV \in \mathcal{K}^{\triangleright}$ . On the other hand, there is no reason to expect that  $\mathcal{K}^{\triangleright}$  is closed with respect to direct sums. Hence it need not be matrix convex.

A subset  $\mathcal{Y}$  of  $\mathbb{S}^g$  is *contractively stable* if  $\sum C_j^* Y C_j \in \mathcal{Y}$  for all  $Y \in \mathcal{Y}$  whenever  $\sum C_j^* C_j \leq I$ . In general, contractively stable sets need not be levelwise convex, as Example 8.7 shows.

# **Proposition 7.10.** *The set* $\mathcal{K}^{\triangleright}$ *is contractively stable.*

*Proof.* Suppose  $B \in \mathcal{K}^{\triangleright}(m)$ . Let  $n \times m$  matrices  $C_1, \ldots, C_{\ell}$  be such that  $\sum C_k^* C_k \preceq I$  and consider the  $n \times n$  matrix  $D = \sum C_k B C_k^*$ .

Given  $Y \in \mathcal{K}(p)$ , there exists a positive semidefinite  $p \times p$  matrix T of trace at most one such that

$$I\otimes T-\sum B_j\otimes Y_j\succeq 0.$$

Thus,

$$I \otimes T - \sum_{j=1}^{g} D_j \otimes Y_j = I \otimes T - \sum_{j=1}^{g} \sum_k C_k^* B_j C_k \otimes Y_j$$
  
=  $\left(I - \sum C_k^* C_k\right) \otimes T + \sum_k (C_k \otimes I)^* \left(I \otimes T - \sum_j B_j \otimes Y_j\right) (C_k \otimes I) \succeq 0.$ 

Hence  $D \in \mathcal{K}^{\triangleright}$ , and the proof is complete.

The *contractive convex hull* of  $\mathcal{Y}$  is the smallest levelwise closed set containing  $\mathcal{Y}$  that is contractively stable. The following proposition finds the two hulls defined by applying the two notions of tracial polar duals introduced above.

**Proposition 7.11.** For  $\mathcal{K} \subseteq \mathbb{S}^{g}$ , the set  $\widehat{\mathcal{K}}^{\triangleright}$  is the levelwise closed convex contractively tracial hull of  $\mathcal{K}$ . Similarly,  $(\widehat{\mathcal{K}})^{\triangleright}$  is the levelwise closed contractively stable hull of  $\mathcal{K}$ .

The proof of the second statement rests on the following companion to Lemma 7.4. Recall from (1.5) the opp-tracial spectrahedron,

$$\mathfrak{H}_Y^{\mathrm{opp}} = \Big\{ B : \exists T \succeq 0, \, \mathrm{tr}(T) \leq 1, \, I \otimes T - \sum B_j \otimes Y_j \succeq 0 \Big\}.$$

**Lemma 7.12.** Fix positive integers m, n, and suppose that S is a nonempty subset of  $\mathbb{S}_n^g$ . Let  $\mathcal{U}$  denote the subset of  $\mathbb{S}_m^g$  consisting of all tuples of the form

$$\sum_{\ell=1}^{\mu} C_{\ell}^* B^{\ell} C_{\ell},$$

where each  $C_{\ell}$  is  $n \times m$ , each  $B^{\ell}$  is in S and  $\sum C_{\ell}^* C_{\ell} \preceq I$ .

(1) If  $Y \in \mathbb{S}_m^g$  is in the conventional polar dual of  $\mathcal{U}$ , then there exists a positive semidefinite  $m \times m$  matrix T with trace at most one such that

$$I \otimes T - \sum B_j \otimes Y_j \succeq 0 \quad \text{for every } B \in \mathcal{S}$$

- (2) The tracial spectrahedra  $\mathfrak{H}_{Y}^{\text{opp}}$  are closed and contractively stable.
- (3) If  $\mathcal{K} \subseteq \mathbb{S}^g$  is contractively stable and  $Y \in \mathbb{S}^g_m$  is in the conventional polar dual  $\mathcal{K}(m)^{\circ c}$  of  $\mathcal{K}(m)$ , then  $\mathcal{K} \subseteq \mathfrak{H}^{opp}_V$ .
- (4) If  $\mathcal{K} \subseteq \mathbb{S}^g$  is levelwise closed and convex, and contractively stable, then

$$\mathcal{K} = \bigcap \{\mathfrak{H}_Y^{\text{opp}} : \mathfrak{H}_Y^{\text{opp}} \supseteq \mathcal{K}\} = \bigcap_n \bigcap \{\mathfrak{H}_Y^{\text{opp}} : Y \in \mathcal{K}(n)^{\circ c}\}$$

(5) The levelwise closed and convex contractively stable hull of  $\mathcal{K} \subseteq \mathbb{S}^g$  is

$$\bigcap \{\mathfrak{H}_Y^{\mathrm{opp}} : \mathfrak{H}_Y^{\mathrm{opp}} \supseteq \mathcal{K}\}$$

(6) For  $\mathcal{K} \subseteq \mathbb{S}^g$ , we have  $Y \in \hat{\mathcal{K}}(n)$  if and only if  $\mathcal{K} \subseteq \mathfrak{H}_Y^{\text{opp}}$ .

*Proof.* The proof of (1) is similar to the proof of Lemma 7.4 and is omitted. Likewise, the proof of (2) follows an argument given in the proof of Proposition 7.10.

To prove (3), suppose that  $Y \in \mathcal{K}(m)^{\circ c}$ . Given  $B \in \mathcal{K}(n)$ , an application of (1) with  $S = \{B\}$  produces an  $m \times m$  positive semidefinite matrix T with  $tr(T) \leq 1$  such that  $I \otimes T - \sum B_j \otimes Y_j \geq 0$ . Hence,  $B \in \mathfrak{H}_Y^{\text{opp}}$ .

We now move on to (4). From (3), if  $Y \in \mathcal{K}(m)^{\circ c}$ , then  $\mathcal{K} \subseteq \mathfrak{H}_{Y}^{\circ pp}$ . On the other hand, if  $Y \in \mathbb{S}_{m}^{g}$  and  $\mathcal{K} \subseteq \mathfrak{H}_{Y}^{\circ pp}$ , then, for  $B \in \mathcal{K}(m)$ ,

$$I\otimes T-\sum B_j\otimes Y_j\succeq 0$$

for some positive semidefinite T with trace at most one. In particular, if we set  $e = \sum_{s=1}^{m} e_s \otimes e_s \in \mathbb{R}^m \otimes \mathbb{R}^m$ , then

$$0 \leq \left\langle I \otimes T - \sum B_j \otimes Y_j e, e \right\rangle = \operatorname{tr}(T) - \lambda_Y(B).$$

Hence  $Y \in \mathcal{K}(m)^{\circ c}$ . Continuing with the proof of (4), from (3) we obtain

$$\mathcal{K} \subseteq \bigcap_{n} \bigcap \{\mathfrak{H}_{Y}^{\mathrm{opp}} : Y \in \mathcal{K}(n)^{\circ c}\}.$$

To establish the reverse inclusion, suppose that *C* is not in  $\mathcal{K}(m)$ . Since  $\mathcal{K}(m)$  is assumed to be closed and convex, there exists a  $Y \in \mathcal{K}(m)^{\circ c}$ , the conventional polar dual of  $\mathcal{K}(m)$ 

(so that  $\lambda_Y(\mathcal{K}(m)) \leq 1$ ) with  $\lambda_Y(C) > 1$ . In particular,  $\mathcal{K} \subseteq \mathfrak{H}_Y^{\text{opp}}$ . On the other hand, if *T* is  $m \times m$  and positive semidefinite with trace at most one, then with  $e = \sum e_s \otimes e_s$ ,

$$\left\langle \left(I \otimes T - \sum C_j \otimes Y_j\right)e, e \right\rangle = \operatorname{tr}(T) - \sum_j \operatorname{tr}(C_j Y_j) = \operatorname{tr}(T) - \lambda_Y(C) < 0$$

Hence,  $C \notin \mathfrak{H}_Y^{\text{opp}}$ .

To prove (5), let  $\mathcal{H}$  denote the contractively stable hull of  $\mathcal{K}$ . Let also  $\mathcal{I}$  denote the intersection of all the tracial spectrahedra  $\mathfrak{H}_Y^{\text{opp}}$  such that  $\mathcal{K} \subseteq \mathfrak{H}_Y^{\text{opp}}$ . Evidently  $\mathcal{H} \subseteq \mathcal{I}$ . On the other hand, by (4),

$$\mathcal{K} \subseteq \mathcal{H} \subseteq \mathcal{I} \subseteq \bigcap \{\mathfrak{H}_Y^{\mathrm{opp}} : \mathfrak{H}_Y^{\mathrm{opp}} \supseteq \mathcal{H}\} = \mathcal{H}.$$

Finally, for (6), first suppose  $Y \in \hat{\mathcal{K}}(n)$ . By definition, for each  $B \in \mathcal{K}$  there is a positive semidefinite T of trace at most one such that  $I \otimes T - \sum B_j \otimes Y_j \succeq 0$ . Hence  $\mathcal{K} \subseteq \mathfrak{H}_Y^{\text{opp}}$ . Conversely, if  $B \in \mathfrak{H}_Y^{\text{opp}}$ , then  $I \otimes T - \sum B_j \otimes Y_j \succeq 0$  for some positive semidefinite T of trace at most one depending on B. Thus, if  $\mathcal{K} \subseteq \mathfrak{H}_Y^{\text{opp}}$ , then  $Y \in \hat{\mathcal{K}}(n)$ .

Proof of Proposition 7.11. Since

$$\widehat{\mathcal{K}^{\triangleright}} = \bigcap_{B \in \mathcal{K}^{\triangleright}} \mathfrak{H}_{B} = \bigcap_{\mathcal{K} \subseteq \mathfrak{H}_{B}} \mathfrak{H}_{B},$$

Theorem 7.6(ii) gives the first conclusion of the proposition.

Likewise,

$$(\hat{\mathcal{K}})^{\triangleright} = \{B : \hat{\mathcal{K}} \subseteq \mathfrak{H}_B\} = \bigcap_{Y \in \hat{\mathcal{K}}} \mathfrak{H}_Y^{\text{opp}} = \bigcap \{\mathfrak{H}_Y^{\text{opp}} : \mathfrak{H}_Y^{\text{opp}} \supseteq \mathcal{K}\},\$$

and the term on the right-hand side is, by Lemma 7.12, the closed contractive convex hull of  $\mathcal{K}$ .

#### 7.5. Matrix convex tracial sets and free cones

In this subsection we introduce and study properties of free (convex) cones.

A subset S of  $\mathbb{S}^g$  is a *free cone* if for all positive integers  $m, n, \ell$ , tuples  $T \in S(n)$ and  $n \times m$  matrices  $C_1, \ldots, C_\ell$ , the tuple  $\sum C_i^* T C_i$  is in S(m). The set S is a *free convex cone* if for all positive integers  $m, n, \ell$ , tuples  $T^1, \ldots, T^\ell \in S(n)$  and  $n \times m$ matrices  $C_1, \ldots, C_\ell$ , the tuple  $\sum_i C_i^* T^i C_i$  lies in S(m). Finally, a subset  $\mathcal{Y}$  of  $\mathbb{S}^g$  is a *contractively tracial convex set* if  $\mathcal{Y}$  is contractively tracial and given positive integers  $m, n, \mu$  and  $Y^1, \ldots, Y^\mu \in \mathcal{Y}(m)$  and  $n \times m$  matrices  $C_1, \ldots, C_\mu$  with  $\sum C_j^* C_j \leq I$ , the tuple

$$\sum C_j Y^j C_j^*$$

lies in  $\mathcal{Y}(n)$ . This condition is an analog to matrix convexity of a set containing 0 which we studied earlier in this paper. Surprisingly:

Proposition 7.13. Every contractively tracial convex set is a free convex cone.

For the proof of this proposition we introduce an auxiliary notion and then give a lemma. A subset  $\mathcal{Y}$  of  $\mathbb{S}^g$  is *closed with respect to identical direct sums* if for each  $Y \in \mathcal{Y}$  and positive integer  $\ell$ , the tuple  $I_{\ell} \otimes Y$  is in  $\mathcal{Y}$ .

**Lemma 7.14.** Suppose  $\mathcal{Y} \subseteq \mathbb{S}^{g}$ .

- (1) If  $\mathcal{Y}$  is contractively tracial and closed with respect to identical direct sums, then  $\mathcal{Y}$  is a free cone.
- (2) If  $\mathcal{Y}$  is contractively tracial and closed with respect to direct sums, then  $\mathcal{Y}$  is a free convex cone.
- (3) If Y is a tracial set containing 0 which is levelwise convex and closed with respect to identical direct sums, then each Y(m) is a cone in the ordinary sense.

*Proof.* To prove (1), let  $Y \in \mathcal{Y}(n)$  and let  $\ell$  be a positive integer. Let  $V_k$  denote the block  $1 \times \ell$  row matrices with  $m \times n$  matrix entries with  $I_n$  in the *k*-th position and 0 elsewhere, for  $k = 1, ..., \ell$ . It follows that  $\sum V_k^* V_k = I$ . Since also  $I_\ell \otimes Y$  is in  $\mathcal{Y}$  and  $\mathcal{Y}$  is tracial,

$$\sum V_k(Y \otimes I_\ell) V_k^* = kY \in \mathcal{Y}(n).$$

Now let *m* and  $\ell$  be positive integers and consider  $m \times n$  matrices  $C_1, \ldots, C_\ell$  and  $Y^1, \ldots, Y^\ell \in \mathcal{Y}(n)$ . Choose a positive integer *k* such that each  $D_j = C_j/\sqrt{k}$  has norm at most one. Let  $M_j$  be the block  $1 \times \ell$  row matrix with  $m \times n$  entries with  $D_j$  in the *j*-th position and 0 elsewhere, for  $j = 1, \ldots, \ell$ . It follows that

$$\sum_{j} M_j^* M_j = \operatorname{diag}(D_1^* D_1, \dots, D_\ell^* D_\ell) \leq I.$$

Since  $\mathcal{Y}$  is tracial, either assuming  $Y^j = Y^k$  for all j, k and  $\mathcal{Y}$  is closed under identical direct sums, or assuming that  $\mathcal{Y}$  is closed under direct sums  $\bigoplus_{i=1}^{\ell} Y^j$  is in  $\mathcal{Y}$ , and hence

$$\sum M_j \Big( \bigoplus_{j=1}^{\ell} Y^j \Big) M_j^* = k \sum_{j=1}^{\ell} D_j Y^j D_j^* = \sum C_j Y^y C_j^* \in \mathcal{Y}(n).$$

Thus, in the first case  $\mathcal{Y}$  is a free cone and in the second a free convex cone.

To prove the third statement, note that the argument used to prove the first part of the lemma shows that if  $\mathcal{Y}$  is a tracial set that is closed with respect to identical direct sums and if each  $C_j$  is I, then  $\ell Y = \sum C_j (Y \otimes I_\ell) C_j^*$  is in  $\mathcal{Y}(n)$ . If  $\mathcal{Y}$  is levelwise convex, since also  $0 \in \mathcal{Y}(n)$ , it follows that  $\mathcal{Y}(n)$  is a convex cone.

*Proof of Proposition 7.13.* Fix positive integers *n* and *v*. Let  $Y^1, \ldots, Y^v \in \mathcal{Y}(n)$ . Let  $C_\ell$  denote the inclusion of  $\mathbb{R}^n$  as the  $\ell$ -th coordinate in  $\mathbb{R}^{nv} = \bigoplus_{i=1}^{v} \mathbb{R}^n$ . In particular,  $C_\ell^* C_\ell = I_n$ , and hence  $Z^\ell = C_\ell Y^\ell C_\ell^* \in \mathcal{Y}(nv)$  (based only on  $\mathcal{Y}$  being a tracial set). Now let  $V_\ell$  denote the block  $v \times v$  matrix with  $n \times n$  entries with  $I_n$  in the  $(\ell, \ell)$  position and zeros  $(n \times n$  matrices) elsewhere. Note that  $\sum V_\ell^* V_\ell = I_{nv}$ . Hence,

$$\sum V_{\ell} Z^{\ell} V_{\ell}^* = \operatorname{diag}(Y^1 \dots Y^{\nu}) \in \mathcal{Y}(n\nu).$$

Thus  $\mathcal{Y}$  is closed with respect to identical direct sums. By Lemma 7.14(2),  $\mathcal{Y}$  is a free convex cone.

**Remark 7.15.** If  $\mathcal{Y} \subseteq \mathbb{S}^g$  is a cone and if  $B \in \mathbb{S}_n^g$  is in the polar dual of the set  $\mathcal{U}$  consisting of all tuples  $\sum C_j Y^j C_i^*$  for  $Y^j \in \mathcal{Y}$  and  $C_j$  such that  $\sum C_i^* C_j \leq I$ , then

$$\sum B_j \otimes Y_j \preceq 0$$

for all  $Y \in \mathcal{Y}(m)$ . In particular, the polar dual  $\mathcal{B} = \mathcal{Y}^{\circ}$  of a cone  $\mathcal{Y}$  is a free convex cone.

To prove this assertion, pick  $B \in S_n^g$  in the polar dual of  $\mathcal{U}$ . Fix a positive integer *m*. By Lemma 7.4, there exists a positive semidefinite *T* with trace at most one such that

$$I\otimes T-\sum B_j\otimes Y_j\succeq 0$$

for all  $Y \in \mathcal{Y}(m)$ . Since  $\mathcal{Y}(m)$  is a cone,  $I \otimes T - \sum B_j \otimes t^2 Y_j \succeq 0$  for all real t, and hence

 $-\sum B_j\otimes Y_j\succeq 0.$ 

It follows that

$$-\sum C^* B_j C \otimes Y_j \succeq 0 \tag{7.4}$$

for any C. The conventional polar dual of a set is convex, which implies convex combinations with various  $C_i$  in (7.4) are in  $\mathcal{B}$ . Hence  $\mathcal{B}$  is a free convex cone.

## 8. Examples

The examples referenced in the body of the paper are gathered together in this section. Some of the examples consider the scalar level  $\Gamma(1) \subseteq \mathbb{R}^g$  of a free set  $\Gamma \subseteq \mathbb{S}^g$ .

**Example 8.1.** This example shows that it is not necessarily possible to choose *V* to be an isometry in equation (2.5) of Theorem 2.4 if the boundedness assumption on  $\mathcal{D}_{\mathfrak{L}_B}$  is omitted. Let g = 1, and consider  $\mathfrak{L}_A(x) = 1 + x$ ,  $\mathfrak{L}_B(x) = 1 + 2x$ . In this case,

$$\mathcal{D}_{\mathfrak{L}_B} = \{X : X \succeq -1/2\} \subseteq \mathcal{D}_{\mathfrak{L}_A} = \{X : X \succeq -1\}$$

It is clear that there does not exist a  $\mu$  and an isometry V such that  $A = V^*(I_{\mu} \otimes B)V$ . This example is in fact representative in the sense that if  $\mathfrak{L}_B$  is a monic linear pencil and  $\mathcal{D}_{\mathfrak{L}_B}$  is unbounded, then there is a monic linear pencil  $\mathfrak{L}_A$  with  $\mathcal{D}_{\mathfrak{L}_B} \subseteq \mathcal{D}_{\mathfrak{L}_A}$  for which there does not exist a  $\mu$  and an isometry V such that  $A = V^*(I_{\mu} \otimes B)V$ .

**Example 8.2.** Here is an example of a trace preserving cp map  $\phi : S \to M_2$  with domain an operator system S that does not admit an extension to a trace nonincreasing cp map  $\phi : M_2 \to M_2$ . This phenomenon contrasts with the classical Arveson extension theorem [Arv69] which says that any ucp map extends to the full algebra.

Let

$$S = \text{span}\{I_2, E_{1,2}, E_{2,1}\}, \quad V = \begin{pmatrix} \sqrt{1/2} & 0 \\ 0 & \sqrt{3/2} \end{pmatrix},$$

and consider the cp map  $\phi : S \to M_2$ ,

$$\phi(A) = V^* A V \quad \text{for } A \in \mathcal{S}.$$

We have

$$\phi(I_2) = V^* V = \begin{pmatrix} 1/2 & 0\\ 0 & 3/2 \end{pmatrix}, \quad \phi(E_{1,2}) = \frac{\sqrt{3}}{2} E_{1,2}, \quad \phi(E_{2,1}) = \frac{\sqrt{3}}{2} E_{2,1},$$

so  $\phi$  is trace preserving on S.

Now let us consider a cp extension (still denoted by  $\phi$ ) of  $\phi$  to  $M_2$ . Let

$$\phi(E_{1,1}) = \begin{pmatrix} a & b \\ b & c \end{pmatrix}$$

Then the Choi matrix for  $\phi$  is

$$C = \begin{pmatrix} a & b & 0 & \sqrt{3}/2 \\ b & c & 0 & 0 \\ 0 & 0 & 1/2 - a & -b \\ \sqrt{3}/2 & 0 & -b & 3/2 - c \end{pmatrix} \succeq 0.$$

Suppose  $\phi: M_2 \to M_2$  is trace nonincreasing. Then

$$1 = \operatorname{tr}(E_{1,1}) \ge \operatorname{tr}(\phi(E_{1,1})) = a + c,$$
  
$$1 = \operatorname{tr}(E_{2,2}) \ge \operatorname{tr}(\phi(E_{2,2})) = 2 - a - c,$$

whence a + c = 1. Since C is positive semidefinite, the nonnegativity of the diagonal of C now gives us  $0 \le a \le 1/2$ . But then the 2 × 2 minor

$$\begin{pmatrix} a & \sqrt{3}/2 \\ \sqrt{3}/2 & 1/2 + a \end{pmatrix}$$

p = 1

is not positive semidefinite, a contradiction.

Example 8.3. Consider

$$-x_1^2 - x_2^4$$
.

In this case p is symmetric with p(0) = 1 > 0.



Bent TV screen  $\mathcal{D}_p(1) = \{(x_1, x_2) \in \mathbb{R}^2 : 1 - x_1^2 - x_2^4 \ge 0\}.$ 

The free semialgebraic set  $\mathcal{D}_p$  is called the real *bent free TV screen*, or (bent) TV screen for short. While  $\mathcal{D}_p(1)$  is convex, it is known that  $\mathcal{D}_p$  is not matrix convex (see [DHM07] or [BPR13, Chapter 8]). Indeed, already  $\mathcal{D}_p(2)$  is not a convex set.



A nonconvex 2-dimensional slice of  $\mathcal{D}_p(2)$ .

That the set  $\mathcal{D}_p(1)$  is a spectrahedral shadow is well known. Indeed, letting

$$L(x_1, x_2, y) = \begin{pmatrix} 1 & 0 & x_1 \\ 0 & 1 & y \\ x_1 & y & 1 \end{pmatrix} \oplus \begin{pmatrix} 1 & x_2 \\ x_2 & y \end{pmatrix},$$

we readily check that  $\operatorname{proj}_x \mathcal{D}_L(1) = \mathcal{D}_p(1)$ . Further, Lemma 4.1 implies that *L* can be replaced by a monic linear pencil  $\mathfrak{L}$ . An explicit construction of such an  $\mathfrak{L}$  can be found in [HKM16, §7.1]. We remark that  $\operatorname{proj}_x \mathcal{D}_L$  strictly contains the matrix convex hull of  $\mathcal{D}_p$  [HKM16, §7.1].

The next example is one in a classical commutative situation. We refer the reader to [BPR13] for background on classical convex algebraic geometry.

**Example 8.4** (The polar dual of the bent TV screen  $\mathcal{D}_p = \{(X, Y) : 1 - X^2 - Y^4 \geq 0\}$ ). We note that  $\mathcal{D}_p^{\circ}(1)$  coincides with the classical polar dual of  $\mathcal{D}_p(1)$  by Proposition 4.3 (cf. [HKM16, Example 4.7]).

We first find the boundary  $\partial D_p^{\circ}(1)$  using Lagrange multipliers. Consider a linear function  $1 - (c_1x + c_2y)$  that is nonnegative but not strictly positive on  $D_p^{\circ}(1)$ , and its values on (the boundary of)  $D_p(1)$ . The KarushKuhnTucker (KKT) conditions for first order optimality give

$$1 - x^2 - y^4 = 0$$
,  $c_1 = 2\lambda x$ ,  $c_2 = 4\lambda y^3$ ,  $1 = c_1 x + c_2 y$ .

Eliminating x, y,  $\lambda$  leads to the following formula relating  $c_1$ ,  $c_2$ :

$$q(c_1, c_2) := -16c_1^8 + 48c_1^6 - 48c_1^4 - 8c_1^4c_2^4 + 16c_1^2 - 20c_1^2c_2^4 - c_2^8 + c_2^4 = 0$$

Thus the boundary of  $\mathcal{D}_p^{\circ}(1)$  is contained in the zero set of q. Since q is irreducible,  $\partial \mathcal{D}_p^{\circ}(1)$  in fact equals the zero set of q. In particular,  $\mathcal{D}_p^{\circ}(1) = \{(x, y) \in \mathbb{R}^2 : q(x, y) \ge 0\}$  is not a spectrahedron, since it fails the line test in [HV07].



Polar dual  $\mathcal{D}_p^{\circ}(1)$  of the bent TV screen.

**Example 8.5.** Recall the free bent TV screen is the nonnegativity set  $D_p$  for the polynomial  $p = 1 - x^2 - y^4$  (see Example 8.3). Let  $\mathcal{K}$  denote the closed matrix convex hull of  $D_p$ . Then  $\mathcal{K}(1) = D_p(1)$ , and hence, by Proposition 4.3 and Example 8.4,  $D_p^{\circ}(1) = D_p(1)^{\circ}$  is not a spectrahedron. Consequently,  $D_p^{\circ}$  is not a free spectrahedron. In particular,  $\mathcal{K}$  cannot be represented by a single  $\Omega$  as in Theorem 4.6.

**Example 8.6.** Tracial and contractively tracial hulls need not be convex (levelwise), as this example shows. Consider

$$A = \begin{pmatrix} 1 & 0 \\ 0 & 0 \end{pmatrix}, \quad B = \begin{pmatrix} 0 & 0 \\ 0 & -1 \end{pmatrix}.$$

To show that  $D = \frac{1}{2}(A + B)$  is not in thull( $\{A, B\}$ ), suppose there exist  $2 \times 2$  matrices  $V_1, \ldots, V_m$  such that  $\sum V_i^* V_j = I$  and

$$\sum V_j A V_j^* = D.$$

On the one hand, the trace of *D* is zero, on the other hand,  $\sum V_j A V_j^*$  has trace 1. Hence *D* is not in the tracial hull of *A*. A similar argument shows that *D* is not in the tracial hull of *B*. Hence by Lemma 6.3,  $D \notin$  thull({*A*, *B*}).

Now consider the tuples  $A = (A_1, A_2)$  and  $B = (B_1, B_2)$  defined by

$$A_1 = \begin{pmatrix} 1 & 0 \\ 0 & 0 \end{pmatrix} = -B_2, \quad A_2 = \begin{pmatrix} 0 & 0 \\ 0 & 1 \end{pmatrix} = -B_1.$$

In this case  $D = \frac{1}{2}(A + B)$  is

$$D = (D_1, D_2) = \frac{1}{2} \left( \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix}, \begin{pmatrix} -1 & 0 \\ 0 & 1 \end{pmatrix} \right).$$

Suppose  $\sum C_j^* C_j \leq I$ . Let

$$F_k = \sum C_j A_k C_j^*$$

and note  $tr(F_k) \ge 0$ . On the other hand,  $tr(D_k) = 0$ . Hence, if  $F_k = D_k$ , then

$$0 = \operatorname{tr}(F_k) = \sum_j \operatorname{tr}(C_j A_k C_j^*) \ge 0.$$

But then, for each j,

$$0 = \operatorname{tr}((C_i(A_1 + A_2)C_i^*)) = \operatorname{tr}(C_iC_i^*).$$

It follows that  $C_j = 0$  for each *j*, and thus  $F_k = 0$ , a contradiction. Thus, *D* is not in the contractive tracial hull of *A*, and by symmetry it is not in the contractive tracial hull of *B*. By Lemma 6.4, *D* is not in the contractive tracial hull generated by {*A*, *B*}.

The following example shows that a contractively stable set need not be convex.

**Example 8.7.** Consider the  $2 \times 2$  matrices A, B from Example 8.6. The smallest contractively stable set containing A, B is the levelwise closed set

$$\mathcal{Y} = \left\{ \sum C_j^* A C_j : \sum C_j^* C_j \leq I \right\} \cup \left\{ \sum D_j^* B D_j : \sum D_j^* D_j \leq I \right\}.$$

Each matrix in  $\mathcal{Y}$  is either positive semidefinite or negative semidefinite, so  $\frac{1}{2}(A+B) \notin \mathcal{Y}$ .

Acknowledgments. The authors thank Man-Duen Choi for stimulating discussions and the referees for their valuable suggestions for improving the exposition.

Research of W. Helton was supported by the National Science Foundation (NSF) grant DMS 1201498, and the Ford Motor Co.

I. Klep was supported by the Marsden Fund Council of the Royal Society of New Zealand, and by the Slovenian Research Agency grants P1-0222, L1-4292 and L1-6722. Part of this research was done while the author was on leave from the University of Maribor.

Research of S. McCullough was supported by the NSF grants DMS 1101137 and 1361501.

## References

- [AM15] Agler, J., McCarthy, J. E.: Global holomorphic functions in several non-commuting variables. Canad. J. Math. **67**, 241–285 (2015) Zbl 1311.32001 MR 3314834
- [AG15] Ambrozie, C.-G., Gheondea, A.: An interpolation problem for completely positive maps on matrix algebras: solvability and parametrisation. Linear Multilinear Algebra 63, 826–851 (2015) Zbl 1326.46044 MR 3291567
- [Arv69] Arveson, W.: Subalgebras of C\*-algebras. Acta Math. **123**, 141–224 (1969) Zbl 0194.15701 MR 0253059
- [Arv72] Arveson, W.: Subalgebras of C\*-algebras II. Acta Math. **128**, 271–308 (1972) Zbl 0245.46098 MR 0394232
- [BB07] Ball, J. A., Bolotnikov, V.: Interpolation in the noncommutative Schur–Agler class. J. Operator Theory 58, 83–126 (2007) Zbl 1164.47023 MR 2336046
- [Bar02] Barvinok, A.: A Course in Convexity. Grad. Stud. Math. 54, Amer. Math. Soc. (2002) Zbl 1014.52001 MR 1940576
- [BN02] Ben-Tal, A., Nemirovski, A.: On tractable approximations of uncertain linear matrix inequalities affected by interval uncertainty. SIAM J. Optim. 12, 811–833 (2002) Zbl 1008.90034 MR 1884919
- [BPR13] Blekherman, G., Parrilo, P. A., Thomas, R. R. (eds.): Semidefinite Optimization and Convex Algebraic Geometry. MOS-SIAM Ser. Optim. 13, SIAM (2013) Zbl 1260.90006

- [BCR98] Bochnak, J., Coste, M., Roy, M.-F.: Real Algebraic Geometry. Ergeb. Math. Grenzgeb. 36, Springer (1998) Zbl 0912.14023 MR 1659509
- [dOHMP09] de Oliveira, M., Helton, J. W., McCullough, S., Putinar, M.: Engineering systems and free semi-algebraic geometry. In: Emerging Applications of Algebraic Geometry, Springer, 17–61 (2009) Zbl 1156.14331 MR 2500463
- [DHM07] Dym, H., Helton, J. W., McCullough, S.: Irreducible noncommutative defining polynomials for convex sets have degree four or less. Indiana Univ. Math. J. 56, 1189– 1232 (2007) Zbl 1128.47020 MR 2333470
- [EW97] Effros, E. G., Winkler, S.: Matrix convexity: operator analogues of the bipolar and Hahn–Banach theorems. J. Funct. Anal. 144, 117–152 (1997) Zbl 0897.46046 MR 1430718
- [Far00] Farenick, D. R.: Extremal matrix states on operator systems. J. London Math. Soc. 61, 885–892 (2000) Zbl 0960.46038 MR 1766112
- [GN11] Gouveia, J., Netzer, T.: Positive polynomials and projections of spectrahedra. SIAM J. Optim. 21, 960–976 (2011) Zbl 1231.13021 MR 2837559
- [Ha11] Hardy, L.: Foliable operational structures for general probabilistic theories. In: Deep Beauty, Cambridge Univ. Press, 409–442 (2011) Zbl 1236.81012 MR 2752528
- [HKM12] Helton, J. W., Klep, I., McCullough, S.: The convex Positivstellensatz in a free algebra. Adv. Math. 231, 516–534 (2012) Zbl 1260.14071 MR 2935397
- [HKM13a] Helton, J. W., Klep, I., McCullough, S.: The matricial relaxation of a linear matrix inequality. Math. Program. 138, 401–445 (2013) Zbl 1272.15012 MR 3034812
- [HKM13b] Helton, J. W., Klep, I., McCullough, S.: Free convex algebraic geometry. In: Semidefinite Optimization and Convex Algebraic Geometry, G. Blekherman et al. (eds.), SIAM, 341–405 (2013) MR 3050247
- [HKM16] Helton, J. W., Klep, I., McCullough, S.: Matrix convex hulls of free semialgebraic sets. Trans. Amer. Math. Soc. 368, 3105–3139 (2016) Zbl 06551055 MR 3451871
- [HKN14] Helton, J. W., Klep, I., Nelson, C.: Noncommutative polynomials nonnegative on a variety intersect a convex set. J. Funct. Anal. 266, 6684–6752 (2014) Zbl 1327.13090 MR 3198852
- [HM04] Helton, J. W., McCullough, S.: A Positivstellensatz for noncommutative polynomials. Trans. Amer. Math. Soc. **356**, 3721–3737 (2004) Zbl 1071.47005 MR 2055751
- [HM12] Helton, J. W., McCullough, S.: Every free basic convex semi-algebraic set has an LMI representation. Ann. of Math. (2) 176, 979–1013 (2012) Zbl 1260.14011 MR 2950768
- [HMPV09] Helton, J. W., McCullough, S., Putinar, M., Vinnikov, V.: Convex matrix inequalities versus linear matrix inequalities. IEEE Trans. Automat. Control 54, 952–964 (2009) MR 2518108
- [HN10] Helton, J. W., Nie, J.: Semidefinite representation of convex sets. Math. Program.
  122, 21–64 (2010) Zbl 1192.90143 MR 2533752
- [HS12] Helton, J. W., Spitkovsky, I. M.: The possible shapes of numerical ranges. Oper. Matrices 6, 607–611 (2012) Zbl 1270.15014 MR 2987030
- [HV07] Helton, J. W., Vinnikov, V.: Linear matrix inequality representation of sets. Comm. Pure Appl. Math. 60, 654–674 (2007) Zbl 1116.15016 MR 2292953
- [Hen10] Henrion, D.: Semidefinite geometry of the numerical range. Electron. J. Linear Algebra 20, 322–332 (2010) Zbl 1205.14076 MR 2679731
- [JKPP11] Johnston, N., Kribs, D. W., Paulsen, V. I., Pereira, R.: Minimal and maximal operator spaces and operator systems in entanglement theory. J. Funct. Anal. 260, 2407–2423 (2011) Zbl 1252.46051 MR 2772376

[KVV14]	Kalyuzhnyi-Verbovetskyi, D., Vinnikov, V.: Foundations of Free Noncommutative Function Theory. Amer. Math. Soc. (2014) Zbl 1312.46003 MR 3244229
[KTT13]	Kellner, K., Theobald, T., Trabandt, C.: Containment problems for polytopes and spectrahedra. SIAM J. Optim. 23, 1000–1020 (2013) Zbl 1296.68177 MR 3055247
[KS13]	Klep, I., Schweighofer, M.: An exact duality theory for semidefinite programming based on sums of squares. Math. Oper. Res. <b>38</b> , 569–590 (2013) Zbl 1309.13031 MR 3092548
[Kls14]	Kleski, C.: Boundary representations and pure completely positive maps. J. Operator Theory <b>71</b> , 45–62 (2014) Zbl 1349.46061 MR 3173052
[Kle07]	Klesse, R.: Approximate quantum error correction, random codes, and quantum chan- nel capacity. Phys. Rev. A <b>75</b> , 062315 (2007)
[Las09]	Lasserre, J. B.: Convex sets with semidefinite representation. Math. Program. <b>120</b> , 457–477 (2009) Zbl 1198.14055 MR 2505746
[LP11]	Li, CK., Poon, YT.: Interpolation by completely positive maps. Linear Multilinear Algebra <b>59</b> , 1159–1170 (2011) Zbl 1236.15004 MR 2837768
[MS11]	Muhly, P. S., Solel, B.: Progress in noncommutative function theory. Sci. China Ser. A <b>54</b> , 2275–2294 (2011)
[Nem06]	Nemirovski, A.: Advances in convex optimization: conic programming. In: Proc. Int. Congress of Mathematicians (Madrid, 2006), Eur. Math. Soc., 413–444 (2007) Zbl 1135.90379 MR 2334199
[NCSB98]	Nielsen, M. A., Caves, C. M., Schumacher, B., Barnum, H.: Information-theoretic approach to quantum error correction and reversible measurement. In: Quantum Coherence and Decoherence (Santa Barbara, CA, 1996), R. Soc. London Proc. Ser. A Math. Phys. Engrg. Sci. <b>454</b> , 277–304 (1998) Zbl 0982.94006 MR 1665553
[NC10]	Nielsen, M. A., Chuang, I. L.: Quantum Computation and Quantum Information. Cambridge Univ. Press (2000) Zbl 1049.81015 MR 1796805
[Par06]	Parrilo, P.: Exact semidefinite representation for genus zero curves. Talk at the Banff workshop, Positive Polynomials and Optimization (Banff, 2006)

[Pau02] Paulsen, V.: Completely Bounded Maps and Operator Algebras. Cambridge Univ. Press (2002) Zbl 1029.47003 MR 1976867

Popescu, G.: Free holomorphic automorphisms of the unit ball of  $B(H)^n$ . J. Reine [Pop10] Angew. Math. 638, 119–168 (2010) Zbl 1196.47005 MR 2595338

- [PNA10] Pironio, S., Navascués, M., Acín, A.: Convergent relaxations of polynomial optimization problems with noncommuting variables. SIAM J. Optim. 20, 2157-2180 (2010) Zbl 1228.90073 MR 2650843
- [Row80] Rowen, L. H.: Polynomial Identities in Ring Theory. Academic Press (1980) Zbl 0461.16001 MR 0576061
- Skelton, R. E., Iwasaki, T., Grigoriadis, K. M.: A Unified Algebraic Approach to [SIG97] Linear Control Design. Taylor & Francis (1997) MR 1484416
- Voiculescu, D.-V.: Free analysis questions I: Duality transform for the coalgebra of [Voi04]  $\partial_{X:B}$ . Int. Math. Res. Notices 2004, 793–822 Zbl 1084.46053 MR 2036956
- [Voi10] Voiculescu, D.-V.: Free analysis questions II: The Grassmannian completion and the series expansions at the origin. J. Reine Angew. Math. 645, 155-236 (2010) Zbl 1228.46058 MR 2673426
- Webster, C., Winkler, S.: The Krein-Milman theorem in operator convexity. Trans. [WW99] Amer. Math. Soc. 351, 307-322 (1999) Zbl 0908.47042 MR 1615970
- Zalar, A.: Operator Positivstellensätze for noncommutative polynomials positive [Zal17] on matrix convex sets. J. Math. Anal. Appl. 445, 32-80 (2017) Zbl 06636763 MR 3543758

workshop, Positive Polynomials and Optimization (Banff, 2006)

### Index

0 is in the interior, 1859 *L*<sub>*A*</sub>, 1847, 1852  $\lambda_B$ , 1878  $\mathfrak{L}_A^{\mathcal{S}}, 1852$  $\mathbb{S}^g, 1846$  $\mathbb{S}_{n}^{g}, 1846$  $\Lambda_A, 1852$ ampliation, 1854 bent free TV screen, 1892 bounded, uniformly, 1851 Choi matrix, 1854 closed with respect to (simultaneous) conjugation by contractions, 1852 closed with respect to conjugation by isometries, 1847 closed with respect to convex direct sums, 1877 closed with respect to direct sums, 1847 closed with respect to identical direct sums, 1889 closed with respect to restriction to reducing subspaces, 1851 completely positive, 1854 contractive convex hull, 1886 contractive tracial hull, 1875, 1877 contractively stable, 1886 contractively tracial, 1849, 1877 contractively tracial convex set, 1888 contractively tracial convex sets, 1846 conventional polar dual, 1878 cthull, 1849, 1877 ex situ tracial dual, 1884 free (noncommutative) matrix-valued polyno-

mial, 1851 free cone, 1888 free convex cone, 1888 free linear matrix inequality (free LMI), 1847 free LMI domain, 1847, 1852 free polar dual, 1859 free polynomial, 1851 free semialgebraic set, 1851 free set, 1851 free spectrahedron, 1847, 1852 free spectrahedrop, 1847, 1859 freely LMI representable, 1852

homogeneous (truly) linear pencil, 1852

in situ , 1886 involution, 1851

length, 1851 levelwise, 1849 levelwise convex, 1849 linear pencil, 1847 LMI domination problem, 1853

matrix (free) convex combination, 1852 matrix convex set, 1847 monic linear pencil, 1847, 1852

nonnegativity set, 1851

operator system, 1854 opp-tracial spectrahedron, 1850, 1886

#### SDP, 1849

SDP representation, 1858 semidefinite programming (SDP) representable set, 1858 spectrahedral shadow, 1858 spectrahedrops, 1847 stratospherically bounded, 1864 symmetric, 1851, 1854

trace nonincreasing, 1857 trace preserving, 1857 tracial, 1875 tracial hull, 1874, 1875 tracial spectrahedra, 1879 tracial spectrahedron, 1849 truncated quadratic module, 1871

uniformly bounded, 1851 unital, 1857 unitary conjugation, simultaneous, 1850